

International Journal of Research Publication and Reviews

Journal homepage: www.ijrpr.com ISSN 2582-7421

2D Nanomaterials in Personalized Medicine from Synthesis to Clinical Translation

Ms A. Nazeer, Ms Atiya Ahmad, F. Ahmad, S. Ahmad

508 BWT, Eros Garden Charmwood Village, Faridabad, Haryana, 121009 drsahmad@email.com

Abstract

Personalized medicine aims to tailor therapeutic strategies to individual patient profiles, enhancing treatment efficacy and minimizing side effects. However, challenges such as heterogeneity of diseases, limited targeting precision, and real-time monitoring hinder its full potential. Two-dimensional (2D) nanomaterials are emerging as promising candidates to overcome these barriers due to their unique physicochemical properties, high surface area, and versatile functionalization capabilities. This review provides a comprehensive overview of the advantages of 2D nanomaterials in personalized medicine, emphasizing their roles in targeted therapy, sensitive diagnostics, and integrated theranostic platforms. We highlight recent advancements where these nanomaterials facilitate selective drug delivery, improve imaging sensitivity, and enable simultaneous diagnosis and treatment monitoring. Despite significant progress, challenges remain in terms of biocompatibility, large-scale synthesis, and precise control over in vivo behavior. Looking forward, interdisciplinary efforts are aiming at optimizing design, improving safety profiles, and establishing regulatory frameworks for successful clinical translation. This review underscores the potential of 2D nanomaterials to revolutionize personalized medicine, advocating for continued research to bridge the gap between laboratory innovation and patient-centered healthcare solutions.

Subject Index: Personalized Medicine; Therapeutic Targeting; Disease Heterogeneity; Two-Dimensional (2D) Nanomaterials; Physicochemical Properties of Nanomaterials; Targeted Drug Delivery; Diagnostic Imaging; Theranostic Platforms; Biocompatibility and Safety; Clinical Translation and Regulatory Challenges

1. Introduction

Conventional "one-size-fits-all" therapeutic regimens have long dominated clinical practice, yet their limitations have become increasingly apparent. Patient responses to standard treatments vary widely: a therapy that proves efficacious in one individual may elicit minimal benefit—or even adverse reactions—in another. This variability stems from inter-individual heterogeneity in genetic makeup, epigenetic modifications, metabolic pathways, and environmental exposures, which together influence disease pathogenesis and drug pharmacokinetics/pharmacodynamics. For example, the advent of targeted oncology agents has revealed that tumor heterogeneity and evolving resistance mechanisms can rapidly undermine treatment efficacy, produce significant clinical setbacks and contribute to treatment failure rates that remain unacceptably high. Moreover, ineffective therapies impose substantial economic burdens on healthcare systems: in the United States alone, estimates suggest that up to 25% of annual drug expenditures yield little to no patient benefit, translating to tens of billions of dollars in wasted resources and exacerbating cost barriers for patients. Collectively, these challenges underscore the imperative for a paradigm shift toward personalized medicine, in which therapeutic strategies are tailored to the unique molecular and physiological profiles of individual patients [Goetz, and Schork,2018].

The discovery of graphene in 2004 catalysed a flurry of research into two-dimensional (2D) nanomaterials, characterized by atomic-scale thickness and extended lateral dimensions. Graphene's landmark isolation revealed exceptional properties—remarkable electrical conductivity, mechanical strength, and high surface area—that spurred exploration of other 2D analogues, including transition metal dichalcogenides (TMDs), black phosphorus, layered double hydroxides (LDHs), MXenes, and covalent organic frameworks (COFs). Over the past decade, these materials have been investigated for diverse biomedical applications. Their unique physicochemical attributes—tunability of surface chemistry, facile functionalization, and intrinsic optical/electronic properties—render them ideal platforms for personalized medicine approaches [Murali, et al, 2021].

High specific surface area enables dense loading of therapeutic cargos (e.g., small molecules, nucleic acids) and targeting ligands, enhancing drug delivery efficiency and enabling controlled release kinetics [Murali, et al, 2021].

Versatile chemical functionalization allows conjugation of biomolecules (antibodies, peptides) for active targeting to diseased tissues, improving therapeutic precision and minimizing off-target effects [Shen, et al, 2012].

Distinct optical/electronic properties facilitate multimodal imaging and real-time monitoring of therapeutic response, integrating diagnostic and therapeutic functions in theranostic platforms called "2D theranostics" [Murali, et al, 2021].

Historically, early biomedical studies focused on graphene oxide (GO) for drug delivery and biosensing, demonstrating improved drug-loading via π - π stacking and electrostatic interactions as well as fluorescence-based cellular imaging. Subsequent work expanded to other 2D families: TMDs for photothermal and photodynamic therapies, MXenes for electrochemical biosensing, and LDHs for pH-responsive drug release in tumor microenvironments. These advancements exemplify how 2D nanomaterials harness structure–function relationships at the nanoscale to enable personalized therapeutic modalities [Wu, et al, 2025].

This review examines the intersection of personalized medicine and 2D nanomaterials, elucidating how these emerging materials address key challenges in targeted therapy, diagnostics, and theranostics, and outlining their path toward clinical translation. Section 2 discusses the design principles and functionalization strategies of 2D nanomaterials for precision drug delivery and molecular targeting. Section 3 reviews advances in 2D-based diagnostic platforms, including biosensors and imaging agents, with emphasis on sensitivity, specificity, and multiplexing capabilities. Section 4 explores integrated theranostic systems that combine therapeutic delivery with real-time monitoring, highlighting recent in vivo studies. Section 5 addresses current limitations—biocompatibility, large-scale synthesis, and in vivo stability—and presents emerging engineering solutions. Section 6 evaluates regulatory considerations and translational pathways, offering perspectives on bridging laboratory innovations with clinical implementation. Through this structured analysis, we aim to provide a comprehensive roadmap for leveraging 2D nanomaterials to realize the full potential of personalized medicine in next-generation healthcare.

2. Synthesis Strategies for Biomedical Applications

2.1 Top-down approaches (exfoliation methods)

Recent advancements in top-down approaches have focused on improving material quality, yield, and scalability. Techniques such as liquid-phase exfoliation using ultrasonication have been refined to produce defect-free nanosheets with controlled thickness and lateral dimensions. Focused Ion Beam (FIB) milling has emerged as a precision top-down technique to fabricate nanoscale structures for biosensing applications, allowing resolution down to a few nanometers. However, controlling size distribution and surface defects remains a challenge, impacting biocompatibility and reproducibility. Researchers have worked toward integrating exfoliation with purification steps to enhance biomedical applicability [Yadav, et al, 2022].

Bottom-up Synthesis (Hydrothermal, Solvothermal)

Bottom-up methods such as hydrothermal and solvothermal synthesis remain popular for achieving highly crystalline, uniform 2D nanomaterials tailored for biomedical functions. The ability to vary reaction parameters allows fine-tuning of morphology and surface chemistry vital for interactions with biological systems. Novel bottom-up strategies include laser pyrolysis and atomic layer deposition (ALD), enabling ultra-thin films and heterostructures with high precision. Despite advances, scalability and cost-effectiveness are ongoing concerns for transitioning to clinical-grade materials [Yadav, et al, 2022; Hassan, et al, 2023].

Green Synthesis Approaches

Green synthesis has gained momentum, employing plant extracts, bacteria, and fungi as eco-friendly reducing and capping agents, enhancing the biocompatibility of 2D nanomaterials. These methods operate at mild temperatures, eliminating toxic reagents and thereby making the materials more suitable for *in vivo* applications. Significant progress has been made in the biosynthesis of materials like graphene oxide, transition metal dichalcogenides, and MXenes. Optimization for large-scale reproducibility and comprehensive biological safety evaluations are in focus [Baig, et al, 2023]

Scalable production methods

Scalability challenges are being met with improvements in chemical vapor deposition (CVD), liquid-phase exfoliation, and atomic layer deposition (ALD). Chemical vapor deposition now can produce wafer-scale, high-quality 2D films with controlled thickness suitable for medical sensor fabrication. Liquid-phase exfoliation offers a cost-effective and eco-friendly route for bulk production without compromising biocompatibility. ALD enables conformal coating of complex substrates for drug delivery platforms. These methods also incorporate inline characterization for quality control facilitating translational biomedical engineering [Yadav, et al, 2022; Hassan, et al, 2023].

2.2 Surface Functionalization and Bioconjugation

Covalent and non-covalent modifications

Latest advances employ a combination of covalent and non-covalent modifications to optimize stability, specificity, and bio-distribution. Covalent functionalization frequently involves PEGylation, carboxylation, and amine-functionalization for stable biomolecule conjugation, ensured through click chemistry and silanization. Non-covalent approaches utilize π - π interactions, hydrogen bonding, and electrostatic forces, preserving intrinsic electronic properties. Emerging methods include bio-orthogonal chemistry for *in vivo* real-time targeting. Functionalization strategies are now tailored to control immune evasion and improve pharmacokinetics [Murali, et al, 2021].

Targeting ligand attachment

Recent reports highlight sophisticated conjugation of targeting ligands such as antibodies, aptamers, peptides, and small molecules enabling cell-specific targeting and enhancing therapeutic index. Dual-ligand and stimulus-responsive systems are explored for combined targeting and controlled release in cancer and infectious disease models. Techniques for site-specific and orientation-controlled ligand attachment have improved receptor binding efficiency and minimized off-target effects[Sanità, et al, 2020].

Biocompatibility enhancement strategies

Enhancement strategies have evolved to include zwitterionic coatings, biomimetic membranes, and scavenging of reactive oxygen species via antioxidants tethered on 2D nanomaterials. PEGylation remains a gold standard, but alternatives like hyaluronic acid and chitosan coatings are gaining attention for targeted delivery and reduced immunogenicity. Surface charge modulation, hydrophibicity/hydrophobicity balance, and the use of endogenous metabolites for coating have been shown to minimize cytotoxicity and prolong circulation times [Murali, et al, 2021].

2.3 Characterization Techniques for Biological Applications

Structural, morphological, and surface analysis

Cutting-edge characterization combines conventional TEM, SEM, AFM with surface-sensitive XPS, Raman spectroscopy, and advanced techniques like synchrotron-based X-ray absorption for detailed compositional and oxidation state analysis. Cryo-electron microscopy (Cryo-EM) is increasingly employed to visualize nano-bio interfaces at near-atomic resolution, invaluable for understanding biofunctionalization success and interaction mechanisms. In situ characterization during synthesis enables real-time monitoring of growth and surface modification. [Yadav, et al, 2022; Baig, et al., 2023].

Biocompatibility assessment methods

Enhanced biocompatibility studies now combine traditional cytotoxicity (MTT, LDH), hemolysis assays, oxidative stress markers, and genotoxicity with high-content imaging and multiplexed cytokine profiling for immunotoxicity. 3D cell culture and organ-on-chip platforms are used for more physiologically relevant testing. High-throughput screening approaches integrating multi-omics and AI-driven analyses have improved prediction and understanding of nanomaterial—biological system interactions[Sanità, et al, 2020].

In vitro and in vivo evaluation protocols

Recent protocols incorporate multimodal imaging (fluorescence, MRI, PET) for tracking biodistribution and clearance in small animal models. Genetic knockout and humanized models are applied to study immune response pathways in detail. Advanced histological, biochemical, and metabolomic endpoints are combined to comprehensively evaluate safety and therapeutic efficacy. Regulatory-compliant Good Laboratory Practice (GLP) guidelines have been tailored for nanomaterial testing to facilitate translational biomedical applications [Murali, et al, 2021].

3. Categories of 2D Nanomaterials in Personalized Medicine

Two-dimensional (2D) nanomaterials—owing to their high surface-to-volume ratios, tunable physicochemical properties, and facile functionalization—have rapidly advanced personalized medicine between 2020–2025 across drug delivery, diagnostics, and theranostics. Graphene derivatives, transition metal dichalcogenides (TMDs), MXenes, black phosphorus (BP), and emerging 2D materials (layered double hydroxides, hexagonal boron nitride, metal—organic frameworks, and rare-earth-based nanosheets) each offer distinct mechanisms for targeting, stimuli-responsive behavior, and multimodal imaging, paving the way for highly specific, controlled, and integrated therapeutic platforms.

3.1 Graphene-Based Materials

3.1.1 Graphene, Graphene Oxide (GO), and Reduced GO (rGO)

Graphene's single-atom-thick lattice provides exceptional mechanical strength, electrical conductivity, and surface area. GO and rGO introduce oxygen-containing groups (epoxide, hydroxyl, carboxyl) that enhance aqueous dispersibility and enable covalent/noncovalent drug conjugation.

3.1.2 Functionalization Strategies for Targeting

Covalent attachment of polymers (PEG, hyaluronic acid, chitosan) and targeting ligands (folic acid, peptides, antibodies) improves solubility, reduces off-target toxicity, and enables receptor-mediated endocytosis. Charge-conversional GO platforms exploit tumor acidity (pH ~6.8) to switch surface charge and trigger payload release in situ. Redox-responsive linkers (disulfide bonds) harness elevated intracellular glutathione (GSH) levels (1–10 mM) for selective drug unloading [Sattari, et al, 2021].

3.1.3 Drug Loading and Release Mechanisms

 π - π stacking and hydrophobic interactions on GO surfaces achieve high loading efficiencies (up to 90%). Stimuli-responsive release includes the followings.

- pH-triggered bond cleavage accelerates drug detachment in acidic tumor microenvironments (pH 6.5–7.2) [AbouAitah, et al, 2023].
- NIR-induced photothermal heating (808 nm) modulates carrier solubility and promotes on-demand release [Khakpour, et al, 2023].
- Electro-responsive GO devices allow remote control of dual drug (aspirin, doxorubicin) delivery via low voltage stimulation [Khakpour, et al, 2023].

3.2 Transition Metal Dichalcogenides (TMDs)

3.2.1 MoS₂, WS₂, MoSe₂ Applications

MoS₂ nanosheets exhibit strong NIR absorbance and high drug loading capacity, enabling photothermal therapy (PTT) synergized with chemotherapy (e.g., Tegafur hydrogels for skin carcinoma). WS₂ doped with Gd³⁺ offers trimodal imaging (PA/MR/CT) and combined PTT-radiotherapy. MoSe₂ quantum dots serve in pH-responsive chemotherapeutic release and photoacoustic imaging [Campos, et al, 2024]. 3.2.2 Photothermal and Photodynamic Properties

 MoS_2 converts 810–808 nm light into localized heat (PTCE ~50 °C) with controlled irradiation parameters (90–180 J/cm²) providing selective tumor ablation (28% viability in A-431 vs. 78% in fibroblasts). [Campos, et al, 2024]. TMD hybrids (e.g., BSA-WS₂@MB) integrate photosensitizers for combined PTT-PDT under 808 nm lasers, boosting reactive oxygen species (ROS) generation [Fusco, et al, 2020].

3.2.3 Multimodal Theranostic Platforms

Layered nanoparticles combining TMDs with iron oxide (Fe_3O_4) or gadolinium enable magnetic resonance, photoacoustic, CT, and fluorescence imaging alongside PTT/CPT. $MoS_2@Fe_3O_4$ -ICG/Pt(IV) nanoflowers support triple therapy (PTT/PDT/chemo) with enhanced tumor selectivity [Fusco, et al, 2020].

3.3 MXenes: The Rising Stars

3.3.1 Ti₃C₂T_x and Other MXene Variants

 $Ti_3C_2T_X$ exhibits metallic conductivity and hydrophilicity after surface termination (-OH, -F). Its 2D morphology and high electron mobility facilitate biosensor electrode modifications. Physically uniform nanoparticles (50 nm) show superior photothermal conversion ($\eta = 40.2\%$ vs. 14.2%) for cancer PTT [Mathew, and Rout,2021].

3.3.2 Conductivity for Biosensing Applications

MXene-based electrochemical sensors detect biomarkers (tPSA, miRNAs, BCR-ABL fusion gene) with femtomolar limits of detection (0.05fM) and rapid response times (10 min-2 h). Integration with AuNPs, PEDOT:PSS, and MOFs enhances sensitivity and specificity for point-of-care diagnostics [Sengupta, and Hussain, 2025].

3.3.3 Photothermal Therapy and Imaging

RGD-functionalized MXene nanoparticles target integrin $\alpha_v \beta_3$ to localize PTT at 808 nm, eradicating tumors while preserving healthy tissue. MXene-Au nanorod hybrids and hollow hydrogels enable NIR-controlled protein/drug release for wound healing and deep chronic wound therapy via dual photothermal-magnetic stimuli [Kim, et al, 2025].

3.4 Black Phosphorus and Phosphorene

3.4.1 Biodegradability Advantages

BP degrades into non-toxic phosphates and phosphonates under physiological conditions, reducing long-term toxicity. BP nanosheets (BPNSs) and BP quantum dots (BPQDs) show negligible accumulation in major organs, with excretion via urine/feces within days[Zhang, et al, 2023].

3.4.2 NIR Absorption Properties

BP's tunable direct bandgap (0.3–2.0 eV) supports strong absorption across NIR-I/II windows. BP@hydrogel platforms exploit 808 nm irradiation to soften agarose matrices, enabling reversible, on-demand drug release and hydrogel degradation post-treatment. BP@Cu complexes capture Cu²⁺ to accelerate biodegradation while stabilizing PTT efficacy and enabling PET imaging with ⁶⁴Cu labelling [Hu, et al, **2020**].

3.4.3 Combination Therapy Applications

- BP-oxaliplatin hybrid nanosheets combine chemotherapeutic self-stabilization with PTT, camouflaged by stem cell membranes for enhanced tumor targeting (MSCs-BP/DACHPt) [Cao, et al, 2021].
- Niosome-coated BP loaded with ICG and DOX achieves pH- and light-dual-responsive chemo-phototherapy, showing >90% loading and controlled release for DNA damage and PTT/PDT synergy in 3D spheroids [Kumar, et al, 2025].

3.5 Other Emerging 2D Materials

3.5.1 Layered Double Hydroxides (LDHs)

MgAl-LDH nanoadjuvants co-load antigens, adjuvants, and drugs to potentiate immune responses. Sericin-tagged LDH delivering pemetrexed and ZnO QDs offers sustained chemotherapy and imaging for breast cancer, achieving higher cytotoxicity than free drug. LDH scaffolds in tissue engineering facilitate pH-responsive gene/drug release and promote cell uptake in chronic wound healing [Yang, et al, 2022].

3.5.2 Hexagonal Boron Nitride (h-BN)

BNNSs, BN QDs, and BN dots serve as insulating, piezoelectric nanocarriers. BNNS-polymer composites enhance mechanical strength and hydroxyapatite formation for bone repair and controlled antibiotic delivery. BN dots enable high-brightness probes for intracellular gene detection with minimal quenching [Dubey, et al, 2025].

3.5.3 2D Metal-Organic Frameworks (MOFs)

Porous 2D MOFs (MIL-101(Cr), UiO-66(Zr), ZIF-8(Zn), MIL-53(Fe)) achieve drug loading up to 1.4 g/g, tunable release over days, and co-delivery of synergistic drug combinations (e.g., AZT-TP/lamivudine for HIV). Surface functionalization with polymers and proteins controls stability and targeting, enabling stimuli-responsive release and multimodal imaging (MRI, fluorescence) [Khafaga, et al, 2024].

3.5.4 Rare Earth-Based 2D Materials

Layered rare earth hydroxides (LREHs) and doped perovskites exfoliate into flexible nanosheets for drug delivery and T1/T2 MRI contrast. Lanthanide oxyiodide (LaOI) nanosheets offer ultrahigh DOX loading (300wt%), pH-responsive release, and enhanced CT contrast at low doses. Ln₆₀ imine nanocages (Eu₆₀, Tb₆₀) encapsulate DOX for luminescence-guided delivery and acid-triggered release in TNBC models, outperforming free DOX with negligible systemic toxicity [Bai, et al, 2024].

From graphene derivatives to MXenes and emerging rare-earth nanosheets, 2D materials have diversified the toolkit for personalized medicine between 2020–2025. Their modular surfaces enable precise targeting, high-capacity drug loading, and multi-stimuli responsiveness, while intrinsic optical, magnetic, or electrical properties support integrated diagnostics and therapy. Continued innovations in material synthesis, surface engineering, and biointerface design are expected to accelerate clinical translation of these 2D platforms for tailored patient treatments.

4. Precision Drug Delivery Systems

4.1 Patient-specific dosing and targeting

Latest advances in precision drug delivery focus on tailoring drug doses and directing therapies precisely to disease sites for individual patients. Nanotechnology and advanced biomaterials allow for targeting specific tissues or cellular receptors, reducing side effects and enhancing efficacy. Technologies like XTPL's Ultra Precise Dispensing enable ultra-precise drug deposition suitable for personalized regimens. Integration of pharmacogenomics and patient data helps optimize dosing for varied metabolism and genetic backgrounds, improving outcomes while minimizing toxicity.

Stimuli-responsive release mechanisms

Recent drug delivery systems are increasingly designed to respond to specific stimuli—such as pH, enzymes, temperature, or external triggers like light and ultrasound—for controlled drug release. These "smart" systems release drugs only when needed at the disease site, enhancing treatment precision. For example, tumor microenvironment-responsive nanoparticles release chemotherapeutics preferentially in acidic or enzyme-rich cancerous tissue microenvironments.

Overcoming drug resistance

Emerging delivery platforms are engineered to bypass or counteract drug resistance mechanisms, such as efflux pumps or drug degradation. Multifunctional nanoparticles co-delivering drugs with resistance inhibitors, or gene-silencing agents via RNA interference, are being tested. Such platforms enable renewed efficacy against cancers and infectious diseases resistant to standard therapies.

Case studies in cancer therapy

Clinical and preclinical case studies demonstrate the success of personalized drug delivery in cancers such as breast, lung, and glioblastoma, improving survival and reducing side effects by adapting dose schedules and delivery vehicle design to tumor biology and patient genetics [BLOG-01].

4.2 Personalized Diagnostics and Biosensing

${\bf Liquid\ biopsy\ and\ circulating\ biomarkers}$

Liquid biopsy technology has advanced rapidly, allowing non-invasive monitoring of circulating tumor cells (CTCs), circulating tumor DNA (ctDNA), extracellular vesicles, and microRNAs from blood or other body fluids. This approach provides dynamic, real-time insights into tumor heterogeneity, treatment response, and early detection of recurrence, overcoming limitations of invasive tissue biopsies. Multi-analyte liquid biopsies improve diagnostic accuracy and enable personalized treatment monitoring especially in cancers like glioblastoma and breast cancer [Bartolomucci, et al, 2025].

Point-of-care diagnostic devices

Development of portable, easy-to-use diagnostic tools enables rapid testing at the patient bedside or in low-resource settings. These devices incorporate biosensors and microfluidics for sensitive detection of personalized biomarkers, facilitating timely clinical decisions and personalized treatment adjustments.

Wearable and implantable sensors

Wearables and implantable biosensors monitor physiological parameters and biomarkers continuously, offering personalized health monitoring. Integration with AI allows early detection of disease flare-ups or treatment side effects, and real-time drug level monitoring for adaptive therapy adjustments.

AI-enhanced diagnostic platforms

Artificial intelligence (AI) algorithms analyze large datasets from imaging, liquid biopsies, and sensor outputs to improve early diagnosis, predict treatment response, and personalize patient management. AI supports integration of multimodal data for holistic, individualized diagnostics.

4.3 Theranostic Platforms for Individual Patients

· Simultaneous imaging and therapy

Theranostics combines diagnostic imaging agents and therapeutic compounds in a single platform, enabling simultaneous disease visualization and treatment delivery. Examples include radiolabeled nanoparticles that allow tracking of drug delivery and therapeutic impact in cancers and inflammatory diseases with real-time feedback.

• Real-time treatment monitoring

Advanced theranostic systems incorporate sensors or imaging modalities to continuously monitor therapeutic effects, providing immediate data to clinicians for adaptive interventions.

Adaptive therapy protocols

Data from theranostic platforms feed into adaptive treatment regimens that adjust dosing and modality combinations based on real-time patient response, optimizing therapeutic outcomes and minimizing side effects.

4.4 Immunotherapy and Vaccine Delivery

Personalized cancer vaccines

Advances in neoantigen identification and mRNA vaccine technology have accelerated development of personalized cancer vaccines that prime the immune system against patient-specific tumor antigens. These vaccines are tailored through sequencing tumor mutations and synthesizing matching epitopes.

• Immune system modulation

Novel delivery systems enhance immune modulation by precise delivery of checkpoint inhibitors, cytokines, or adjuvants to immune cells or tumor microenvironments, improving efficacy and reducing systemic immune-related adverse effects.

• Antigen presentation enhancement

Nanoparticles and other carriers improve antigen delivery to dendritic cells and other antigen-presenting cells, boosting immune recognition and activation for enhanced vaccine response and immunotherapy durability.

4.5 Regenerative Medicine Applications

Tissue engineering scaffolds

3D-printed and bioengineered scaffolds with patient-specific architecture and biochemical cues support tissue regeneration. Smart scaffolds release growth factors or recruit stem cells with controlled kinetics, tailored to individual healing requirements.

• Stem cell therapy enhancement

Delivery systems improve stem cell viability, homing, and differentiation through co-delivery of supportive factors and protective microenvironments, addressing patient-specific deficits in regenerative capacity.

· Wound healing and repair

Personalized dressings and delivery platforms release antimicrobials, growth factors, and cells to optimize wound healing, particularly in chronic wounds and diabetic ulcers, enhancing repair outcomes based on patient-specific wound conditions.

These developments reflect the integration of cutting-edge nanotechnology, biotechnology, AI, and digital health tools underpinning personalized medicine's evolution during 2020-2025. They enable highly tailored treatment and diagnostic strategies that improve efficacy, reduce side effects, and support adaptive medicine customized to individual patient biology and disease dynamics [de Lima, et al, 2025].

5. Integration with Digital Health Technologies

The convergence of artificial intelligence, telemedicine, and advanced computational modeling is fundamentally reshaping healthcare delivery through sophisticated digital health technologies. This transformation spans personalized treatment algorithms, remote monitoring capabilities, and virtual patient modeling systems that collectively enable unprecedented precision in medical care.

5.1 AI and Machine Learning Integration

Personalized Treatment Algorithms

The development of personalized treatment algorithms represents a paradigm shift from standardized medical care toward individualized therapeutic approaches. AI-driven systems now analyze vast datasets encompassing genomic profiles, medical histories, lifestyle factors, and real-time physiological data to generate patient-specific treatment recommendations[Bongurala, et al, 2025].

Recent advancements demonstrate the clinical efficacy of machine learning models in treatment personalization. For diabetes management, sophisticated algorithms incorporating decision trees, neural networks, and reinforcement learning have shown remarkable success in predicting blood glucose levels and optimizing insulin dosages. These systems leverage data from electronic health records (EHRs), continuous glucose monitoring (CGM) equipment, and wearable devices to deliver real-time treatment adjustments tailored to individual patient physiology [Tanveer, et al, 2025].

In oncology, AI-powered precision medicine has achieved significant breakthroughs in treatment customization. Machine learning algorithms analyze patient genetic profiles, tumor characteristics, and treatment response patterns to generate highly specific therapeutic recommendations. This precision approach maximizes treatment efficacy while minimizing adverse effects, representing a substantial improvement over traditional one-size-fits-all protocols. The integration of AI-guided personalized medicine has demonstrated improved survival rates and enhanced quality of life for cancer patients [Bongurala, et al, 2025]

The genomic considerations in therapy planning have become increasingly sophisticated through AI integration. Machine learning algorithms predict which patients require altered prescribing or dosing based on pharmacogenomically actionable variants. These systems enable real-time recommendations by identifying patients likely to need specific medications before genomic information becomes clinically necessary. This proactive approach has proven particularly valuable in medulloblastoma treatment, where AI-mediated analysis of hundreds of exomes has facilitated the administration of targeted therapies to specific pediatric patient cohorts. [Johnson, et al, 2021].

Predictive Modeling for Therapy Outcomes

Predictive analytics powered by machine learning techniques have revolutionized treatment outcome forecasting across multiple medical specialties. These systems analyze electronic health records, imaging data, and genetic information to predict disease progression, treatment response, and recovery rates with unprecedented accuracy [Dixon, et al, 2024].

Recent studies demonstrate the transformative impact of AI predictive models on patient outcomes. Machine learning algorithms analyzing acute kidney injury (AKI) patients in intensive care units have shown superior performance compared to traditional regression models in predicting renal function recovery and reversibility. Similarly, pediatric traumatic brain injury outcomes can now be predicted with high sensitivity using machine learning algorithms, enabling early prognostic counseling and intervention planning [Dixon, et al, 2024].

The PARAllel predictive MOdeling (PARAMO) platform exemplifies advanced healthcare analytics systems that utilize EHRs to optimize predictive modeling across patient cohorts. This innovative software significantly accelerates computational modeling tasks through parallel processing, enabling rapid analysis of large patient datasets for future disease prediction. Such platforms address vulnerability concerns including data privacy and algorithmic bias while improving personalized care delivery and disease monitoring capabilities [Dixon, et al, 2024].

Immunotherapy outcome prediction has witnessed remarkable advancement through AI-powered histopathological analysis. Machine learning algorithms can now accurately measure immunohistochemical labeling results and identify cancer cells on histology slides with greater precision than human pathologists. AI-powered analyzers based on PD-L1 tumor ratio scores have demonstrated superior performance in predicting immunotherapy success compared to manual pathologist diagnosis in non-small cell lung cancer patients [Dixon, et al, 2024].

Drug Discovery Acceleration

The AI-driven drug discovery sector has experienced explosive growth, with the global market expanding from \$1.5 billion in 2023 to a projected \$20.30 billion by 2030, representing a remarkable 29.7% compound annual growth rate. This unprecedented expansion reflects the increasing adoption of machine learning and deep learning technologies across all stages of pharmaceutical development [MR-01].

Recent innovations demonstrate AI's transformative impact on drug discovery timelines. Virtual screening methodologies utilizing machine learning algorithms can rapidly evaluate vast chemical libraries, dramatically reducing the time required to identify promising drug candidates. These systems employ sophisticated predictive modeling to forecast drug efficacy and toxicity profiles, enabling researchers to prioritize compounds with optimal therapeutic potential while minimizing adverse effects [Serrano, et al, 2024].

The COVID-19 pandemic significantly accelerated AI adoption in drug discovery, particularly from 2020 to 2022. Notable achievements include Exscientia's development of the world's first AI-designed molecule for immuno-oncology, which entered Phase I trials in 2021. This milestone achievement led to a \$100 million Series C funding round, demonstrating investor confidence in AI-driven pharmaceutical development [MR-01].

Multi-omics data integration represents a cutting-edge approach combining genomics, proteomics, and metabolomics data through AI algorithms. This comprehensive analysis enables the identification of novel therapeutic targets and facilitates personalized treatment development based on individual patient molecular profiles. Machine learning algorithms can analyze diverse patient datasets to provide tailored treatments minimizing adverse effects while optimizing therapeutic outcomes [Serrano, et al, 2024].

5.2 Telemedicine and Remote Monitoring

Wearable 2D Material-Based Devices

The development of 2D materials-based wearable sensors has revolutionized remote health monitoring capabilities through unprecedented flexibility, sensitivity, and biocompatibility. These advanced materials, including graphene, transition metal dichalcogenides (TMDCs), hexagonal boron nitride (h-BN), and MXenes, enable skin-integrated devices that provide continuous physiological monitoring with remarkable accuracy [Wang, et al, 2022]. Recent innovations in 2D materials technology have enabled wearable sensors capable of monitoring multiple health parameters simultaneously. These devices can detect body temperature, arterial oxygen saturation, breath rate, muscle fatigue, and vocal cord vibrations while maintaining flexibility and breathability compatible with human skin movements. The exceptional electro-mechanical characteristics of 2D materials make them ideal candidates for sensing various bio-signals, electrocardiograms, and movement patterns[Vaghasiya, et al, 2023].

Graphene-based wearable biosensors have achieved remarkable sensitivity in detecting biomarkers associated with various medical conditions, including lung, liver, pancreatic, and breast cancers. The successful implementation of graphene in biosensing applications has stimulated research into similar materials such as MXenes, metal phosphorus chalcogenides, and black phosphorous, expanding the range of available sensing capabilities[Vaghasiya, et al, 2023].

The RemoteHealthConnect system exemplifies advanced integration of wearable technology with clinical decision-making platforms. This innovative web-based healthcare system utilizes the Vitaliti™ continuous patient monitoring wearable, which simultaneously monitors five critical vital signs: blood pressure, heart rate, body temperature, blood oxygen saturation, and respiratory rate. The system captures and transmits raw physiological signals including ECG, PPG, and accelerometer data in real-time, providing healthcare professionals with comprehensive patient monitoring capabilities [Arun, et al, 2024].

Real-Time Health Parameter Tracking

Advanced data visualization techniques have transformed how healthcare professionals interpret real-time patient data from wearable devices. Custom visualizations enable early detection of health issues, effective remote patient monitoring, and improved healthcare management through intuitive data presentation. The RemoteHealthConnect system has achieved a System Usability Scale (SUS) score of 71.5, positioning it above average usability thresholds for healthcare technologies [Arun, et al, 2024].

The global remote patient monitoring market has experienced substantial growth, valued at \$22.03 billion in 2024 and projected to reach \$110.71 billion by 2033. This expansion reflects increasing adoption of AI-enhanced remote monitoring systems that provide continuous, real-time analysis of biometric data from connected wearables including blood pressure cuffs, pulse oximeters, and glucose monitors [RPMS-01].

AI enhancement of remote patient monitoring delivers predictive insights and personalized care recommendations, enabling clinicians to detect patient deterioration early while reducing alert fatigue. AI-powered systems streamline data processing, triage, and electronic health record integration, facilitating scalable, cost-effective home-based care models. These capabilities become increasingly valuable as remote monitoring becomes integral to chronic disease management, telehealth programs, and hospital-at-home initiatives [RPMS-01].

Recent innovations include the Pylo GL1-LTE cellular connected blood glucose meter, launched by Prevounce Health in June 2024. This clinically validated device securely transmits real-time glucose readings via multiple cellular networks with laboratory-level accuracy, integrating seamlessly with healthcare platforms and systems [RPMS-01].

Home-Based Diagnostic Systems

The at-home testing market has experienced remarkable growth, with global sales estimated at \$7.789 billion in 2025 and projected to reach \$11.878 billion by 2035, representing a 4.8% compound annual growth rate. This expansion reflects increasing consumer preference for self-diagnostic tools and technological advancements improving test accuracy and reliability.[FMI-01].

Digital monitoring systems dominate the home diagnostics market, accounting for 48.2% of global market share in 2025. These systems utilize advanced sensors providing precise data for regular and emergency monitoring, with automatic data storage ensuring accurate record-keeping and regulatory compliance. The integration of predictive capabilities improves maintenance and operational efficiency while reducing overall healthcare costs [FMI-01].

Point-of-care diagnostic testing has revolutionized healthcare accessibility by providing immediate, actionable results at the patient's location. These systems address traditional laboratory limitations including lengthy turnaround times, high costs, and accessibility constraints. Advanced point-of-care devices now support multipurpose testing capabilities, identifying multiple disease types through easily upgradable electronic platforms [POCS-01].

Recent technological advances include AI-enhanced home diagnostic systems that analyze medical images from MRIs and X-rays to detect abnormalities. Wearable devices integrated with AI technologies enable remote monitoring of patient vitals, providing real-time insights to patients and healthcare professionals while facilitating timely interventions. Machine learning algorithms analyze vast patient datasets including medical history and genetic factors to create personalized treatment approaches [MR-02].

The home diagnostics market benefits from expanding applications, compact designs, cost-effectiveness, increased consumer health awareness, and accelerated regulatory approvals. Technological innovations focus on developing more reliable, precise, and sensitive diagnostic kits that are user-friendly and integrated with digital technologies [MR-02].

5.3 Digital Twins and Virtual Patient Models

Personalized Treatment Simulation

Digital twin technology in healthcare creates comprehensive virtual replicas of individual patients by integrating health data, medical history, lifestyle information, and genetic profiles. These sophisticated models enable physicians to simulate various treatment scenarios, compare outcomes, and optimize therapeutic interventions before implementation, fundamentally transforming personalized medicine delivery [Demuth, et al, 2025].

The synergistic integration of digital twins with personalized medicine represents a revolutionary advancement in healthcare conception and delivery. Digital twins provide technological frameworks for continuous treatment modeling and optimization, while personalized medicine focuses on tailoring care based on individual genetic profiles, environmental factors, and lifestyle characteristics. This combination creates dynamic feedback systems enabling data-driven, patient-specific medical decisions that evolve with new information availability [Saratkar, et al, 2025].

Oncological applications demonstrate significant potential for digital twin technology in treatment simulation. Oncologists can create digital twins to simulate tumor behavior and treatment effectiveness prior to therapy administration, utilizing imaging data, genetic mutations, and previous treatment responses. This approach enables dynamic treatment strategy adjustments based on disease progression, resulting in more personalized and less invasive interventions with improved patient outcomes [Saratkar, et al, 2025].

Cardiac digital twins utilize real-time physiological signals from wearable devices including heart rate, blood pressure, and ECG data to predict complications and enable dynamic medication adjustments. Continuous monitoring capabilities allow early identification of potential issues, facilitating proactive intervention and optimized care delivery [Saratkar, et al, 2025].

Dosing Optimization

Model-informed precision dosing (MIPD) has gained significant momentum in oncology, reflecting its potential to revolutionize patient care through individualized pharmacokinetic profile-based treatment tailoring. This approach leverages mathematical population pharmacokinetic models to inform dosing decisions, capturing drug-specific parameters such as clearance and distribution volume while quantifying inter-patient variability [Agema, et al, 2025].

Recent prospective validation studies have identified 16 different oncological drugs for which model-informed precision dosing implementation has been performed. These investigations primarily focus on achieving adequate drug exposures and reducing inter-individual variability, with demonstrated clinical outcome improvements for busulfan and high-dose methotrexate treatments. Significant toxicity reductions have been observed for busulfan and cyclophosphamide therapies [Agema, et al, 2025].

Virtual patient modeling for dosing optimization has demonstrated remarkable potential through simulation-based inference approaches. These methodologies generate parameter probability distributions during individual patient fitting, providing alternative parameterizations for real patients

and enabling larger virtual population generation. This approach overcomes traditional limitations of individual patient data analysis while maintaining foundation in real patient characteristics [Marques, and Vale, 2024].

The development of population pharmacokinetic models using virtual patients has shown comparable predictive performance to traditional clinical trial-derived models. External validation using real-world clinical data demonstrates the feasibility of utilizing synthetic data for therapeutic regimen optimization, offering advantages including flexibility, cost-effectiveness, and diverse patient population simulation capabilities [Marques, and Vale, 2024].

Adverse Effect Prediction

AI-powered adverse event prediction represents a critical application of machine learning in clinical decision support systems, enabling proactive intervention and enhanced patient safety. These systems analyze substantial amounts of routinely collected clinical data, particularly from high-risk environments such as postoperative care and intensive care units, to identify patterns preceding adverse events [Oei, et al, 2025].

Machine learning techniques for adverse event prediction leverage various algorithms including random forest and gradient boosting classifiers to develop predictive models. Treatment effect estimation compares predicted outcomes between treatment and control groups, incorporating treatment predictors, confounders, treatment effect modifiers, and outcome risk factors with known effects on treatment outcomes [Fang, et al, 2019].

Virtual patient cohorts provide valuable platforms for investigating bias in machine learning applications for individual treatment effect prediction. These simulated healthcare datasets enable evaluation of model performance across different patient characteristics and outcome scenarios, identifying potential sources of bias that may affect real-world clinical implementation [Fang, et al, 2019].

Recent research demonstrates that direct machine learning application may not adequately address bias in individual treatment effect prediction. Even models achieving 100% accuracy in health outcome prediction can yield biased individual treatment effects, highlighting the need for sophisticated methodological development to advance machine learning capabilities in individualized treatment selection [Fang, et al, 2019].

Digital twin applications in adverse effect prediction extend beyond individual patient modeling to population-level safety assessment. In-silico clinical trials utilizing digital twins can simulate both control and efficacy arms, optimizing patient recruitment and drug protocols while supporting better-powered clinical trials. Companies such as Unlearn.AI are developing digital twin-based trial designs for neurological diseases including Alzheimer's, Parkinson's, and multiple sclerosis[Katsoulakis, et al, 2024].

The integration of digital health technologies represents an unprecedented transformation in healthcare delivery, combining AI-powered personalized treatment algorithms, advanced wearable monitoring systems, and sophisticated virtual patient modeling capabilities. These innovations collectively enable precision medicine approaches that optimize therapeutic outcomes while minimizing adverse effects, fundamentally reshaping the future of medical care through data-driven, individualized treatment strategies.

6. Clinical Translation Challenges and Regulatory Considerations

The clinical translation of nanomedicines represents one of the most complex undertakings in modern healthcare, requiring navigation of multifaceted safety, manufacturing, regulatory, and economic challenges. Despite significant technological advances, the gap between laboratory innovations and clinical applications remains substantial, with numerous nanomedicine candidates failing to progress beyond preclinical stages due to intricate regulatory requirements and economic barriers.

6.1 Biocompatibility and Safety Assessment

Long-term Toxicity Studies

The evaluation of long-term toxicity in nanomedicines has become increasingly sophisticated, with recent developments emphasizing comprehensive assessment protocols that address the unique properties of nanomaterials. Current biocompatibility testing frameworks, guided by ISO/TR 10993-22:2017, require extensive evaluation including cytotoxicity, genotoxicity, carcinogenicity, reproductive toxicity, immunotoxicity, irritation, sensitization, hemocompatibility, systemic toxicity, pyrogenicity, and implantation studies [Kyriakides, et al, 2021].

Recent research demonstrates that nanomaterial toxicity depends critically on physicochemical properties including chemical composition, size, shape, and surface coating. The unique nanoscale characteristics create exceptional surface-to-volume ratios and quantum effects that fundamentally alter biological interactions compared to bulk materials. These properties enable nanomaterials to readily cross biological membranes and enter cells, tissues, and organs, potentially causing adverse health effects through mechanisms distinct from their larger counterparts [Qamar, et al, 2024].

Carbon-based nanomaterials exemplify the complexity of toxicity assessment in nanomedicine applications. Studies demonstrate that smaller carbon nanoparticles exhibit greater oxidative potential than larger particles in alveolar epithelial cells, with graphene showing the highest toxicity (52.24%) among carbon nanomaterials, followed by carbon nanotubes, fullerenes, and carbon nanowires. The toxicity variations correlate with different structural arrangements and aspect ratios, highlighting the need for comprehensive evaluation of each nanomaterial variant. [Qamar, et al, 2024].

The development of standardized in vitro methods has emerged as a priority for accelerating toxicity assessment while reducing reliance on animal studies. Recent advances focus on establishing correlation between physicochemical features of nanomaterials and their specific bio-effects, though challenges remain due to the broad spectrum of different nanomaterial types, varying model systems, lack of standardized protocols, and increased complexity of in vivo testing environments [Qamar, et al, 2024].

Biodegradation Pathways

Recent investigations into nanomaterial biodegradation have revealed sophisticated pathways that vary significantly based on material composition and environmental conditions. The biodegradation of nanomaterials follows complex enzymatic processes, with bacterial strains demonstrating remarkable capabilities in breaking down synthetic compounds into simpler, less toxic molecules [Eshghi, and Jookar Kashi, 2025].

A notable example involves copper nanoparticles biosynthesized by Micrococcus lylae, which demonstrated effective biodegradation of Acid Red 88 dye through metabolic processes. The degradation pathway revealed enzymatic conversion of complex aromatic compounds into simpler molecules including (Z)-1-hydrazono-1,8a-dihydronaphthalen-2(4aH)-one, sodium naphthalene-1-sulfonate, (Z)-6-hydrazono cyclohex-2-enone, and sodium benzenesulfonate. This biodegradation process achieved complete breakdown within 11 hours when copper nanoparticles were combined with bacterial cell-free extract, demonstrating the potential for accelerated biodegradation through nanomaterial-assisted processes [Eshghi, and Jookar Kashi, 2025]. Environmental biodegradation studies have shown that nanoparticles can improve biodegradation rates through various mechanisms, including enhanced bacterial growth and metabolic activity. The integration of nanomaterials with biodegradation processes offers promising approaches for environmental remediation while providing insights into how these materials behave in biological systems. However, careful consideration of degradation products and their potential toxicity remains essential for comprehensive safety assessment [Remya, et al, 2022].

The biodegradation pathway analysis using advanced techniques such as GC-MS and FTIR spectroscopy enables detailed characterization of degradation products and mechanisms. These analytical approaches confirm complete breakdown of target compounds and help identify intermediate metabolites, providing crucial information for safety evaluation and regulatory approval processes [Eshghi, and Jookar Kashi, 2025].

Immune System Interactions

The interaction between nanomaterials and the immune system has become a central focus of biocompatibility assessment, with recent research revealing both beneficial immunomodulatory effects and potential adverse immunotoxic responses. The immune system's response to nanomaterials depends critically on their physicochemical properties, with surface characteristics, size, and composition playing decisive roles in determining biological outcomes [Aljabali, et al, 2024].

Nanomaterial-driven precision immunomodulation represents an emerging therapeutic strategy that leverages controlled immune system interactions. Liposomes, polymers, and inorganic nanoparticles serve as versatile platforms capable of delivering immunomodulatory molecules with precise targeting capabilities. These systems enable fine-tuning of immune responses through controlled release kinetics and targeted interactions with specific immune cells, offering therapeutic potential for conditions such as cancer and autoimmune diseases. [Aljabali, et al, 2024].

Recent studies demonstrate that surface properties of nanomaterials significantly influence immune activation. Approximately 50% of investigated nanomaterials induce effects related to immune system activation, with surface-related properties identified as the most critical parameter influencing observed immune effects. Complement activation-related hypersensitivity reactions and adaptive immune response activation represent the most frequent effects reported for lipid-based nanoparticles [Halamoda-Kenzaoui, et al, 2018].

The development of standardized immunotoxicity assessment methods has advanced significantly through the implementation of the human Cell Line Activation Test (h-CLAT) using THP-1 cells. This validated approach evaluates nanomaterial immunotoxicity potential by measuring CD86 and CD54 expression levels as indicators of antigen-presenting cell activation. Silver nanoparticles, silica nanoparticles, and titanium dioxide demonstrate distinct activation patterns that correlate with synthesis methods, hydrodynamic diameters, and crystal types [Nishida, et al, 2024].

Inorganic nanoparticles show particularly high immunotoxic potential, with 70% of investigated materials inducing adverse immune system effects. This elevated risk profile necessitates comprehensive evaluation protocols that address both immediate hypersensitivity reactions and long-term immune system modulation. The identification of prevalent endpoints relevant for immunotoxic potential assessment supports the development of more targeted and efficient evaluation strategies for nanotechnology-based products[Halamoda-Kenzaoui, and Bremer-Hoffmann, 2018].

6.2 Manufacturing and Quality Control

Good Manufacturing Practice (GMP) Compliance

The implementation of Good Manufacturing Practice standards in nanomedicine manufacturing has evolved significantly, with recent regulatory developments emphasizing enhanced quality requirements and compliance frameworks. The revised Schedule M under India's Drugs and Cosmetics Act exemplifies the global trend toward more stringent GMP requirements, with small and medium pharmaceutical manufacturers receiving extended deadlines until December 2025 to comply with enhanced standards [INDBRIEF].

Current GMP regulations for nanomedicines follow established pharmaceutical manufacturing principles while addressing unique challenges posed by nanoscale materials. The FDA's Current Good Manufacturing Practice (cGMP) regulations establish minimum requirements for methods, facilities, and controls used in manufacturing, processing, and packaging of nanomedicine products. These regulations ensure product safety, identity, strength, quality, and purity throughout the manufacturing process.[USFDA]

The complexity of nanomedicine manufacturing requires specialized approaches to traditional GMP compliance elements. Quality control systems must address batch-to-batch variability, particle size distribution consistency, surface modification reproducibility, and stability throughout storage and distribution. Manufacturing facilities require specialized equipment capable of handling nanoscale materials while preventing cross-contamination and ensuring worker safety [Tenchov, et al, 2025].

Regulatory authorities have recognized the need for enhanced manufacturing oversight for nanomedicines, implementing requirements for comprehensive documentation of manufacturing processes, quality control procedures, and facility validation. The integration of advanced analytical techniques for real-time monitoring and quality assessment has become essential for maintaining GMP compliance in nanomedicine production [INDBRIEF].

Batch-to-Batch Consistency

Achieving batch-to-batch consistency in nanomedicine manufacturing represents one of the most significant technical challenges facing the industry. The inherent variability in nanomaterial synthesis processes requires sophisticated control strategies to ensure reproducible physicochemical properties and therapeutic performance across production batches [Costa, and Jornitz, 2024].

Continuous manufacturing approaches have emerged as a promising solution for improving batch consistency in nanoparticle production. Unlike traditional batch processing, continuous manufacturing operates under constant process parameters, producing nanoparticle products with consistent quality regardless of batch size. This approach addresses fundamental limitations of batch processing, including limited control, low throughput, and yield variability that become more pronounced at larger scales [Costa, and Jornitz, 2024].

The implementation of real-time monitoring and control systems has proven essential for maintaining batch consistency. Advanced process analytical technology (PAT) enables continuous monitoring of critical quality attributes including particle size distribution, drug loading, encapsulation efficiency, and surface properties. This real-time feedback allows for immediate process adjustments to maintain product specifications within acceptable ranges [Costa, and Jornitz, 2024].

Automated production systems reduce human error and improve reproducibility in nanomedicine manufacturing. The integration of automated dosing, mixing, temperature control, and purification steps ensures consistent processing conditions across batches. These systems enable seamless reproducibility and provide comprehensive documentation required for regulatory compliance and quality assurance [Costa, and Jornitz, 2024].

Recent advances in turbulent mixing technology facilitate processing at high flow rates while maintaining precision in particle formation. This technology ensures uniform nanoparticle size and quality, safeguarding therapeutic efficacy in clinical applications. The incorporation of real-time analytics into continuous processing frameworks allows for mid-stream adjustments based on ongoing monitoring, crucial for maintaining product integrity and preventing batch failures [Costa, and Jornitz, 2024].

Scalability Challenges

The scalability of nanomedicine manufacturing from laboratory to commercial production presents multifaceted challenges that have limited the translation of promising nanomaterials into clinical applications. Traditional scale-up approaches often fail to maintain the precise control required for nanomaterial properties, resulting in significant changes in material characteristics and performance [Ahmed, 2022].

Top-down and bottom-up manufacturing approaches each present distinct scalability challenges. Top-down methods involving mechanical size reduction through advanced lithography and etching can maintain some control over particle properties but require significant capital investment and energy consumption. Bottom-up approaches building materials molecule by molecule offer greater control over nanoscale properties but face challenges in maintaining uniformity and yield at larger scales [Ahmed, 2022].

The cost implications of nanomedicine scale-up represent a major barrier to commercial viability. Production costs for nanomaterials can be orders of magnitude higher than bulk materials, with gold nanoparticles costing approximately \$80,000 per gram compared to \$50 per gram for raw gold. These cost disparities necessitate innovative manufacturing approaches and economies of scale to achieve commercial feasibility [Tenchov, et al, 2025].

Microfluidic manufacturing platforms have emerged as a promising solution for scalable nanomedicine production. These systems enable continuous, scaled-up synthesis of high-quality nanoparticles with superior performance, easy scalability, and reliability compared to conventional batch reactors. Microfluidic platforms can achieve significantly higher throughput than batch reactors while enabling on-line implementation of surface functionalization and sterilization processes [Gomez, et al, 2014].

Recent developments in flow technology for nanoparticle synthesis address critical scale-up issues by operating manufacturing processes under constant parameters. This approach produces consistent quality products irrespective of batch size, overcoming traditional limitations of batch technology that become problematic as production scales increase. Flow technology offers particular advantages in process safety and quality control, though case-by-case evaluation remains necessary to determine optimal manufacturing approaches[PROD-01].

6.3 Regulatory Framework and Approval Pathways

FDA and EMA Guidelines for Nanomedicines

The regulatory landscape for nanomedicines has evolved substantially, with both the FDA and EMA developing comprehensive guidelines specifically addressing the unique challenges posed by nanotechnology-based medicinal products. The EMA's 2025 Horizon Scanning Report provides detailed guidance on regulatory frameworks, current status of authorized nanomedicines, and emerging regulatory challenges in the European Union [NNOTE-011]

Current FDA guidelines emphasize a case-by-case approach for nanomedicine evaluation, recognizing that these products may not fit traditional regulatory categories. The FDA's guidance document "Drug Products, including Biological Products, that Contain Nanomaterials" (2022) provides specific recommendations for pharmaceutical development, non-clinical studies, and early clinical investigations of nanomedicine products. This guidance addresses the complexity of nanosystems and provides information for block-copolymer micelles, liposome-like products, and nanoparticle iron medicinal drug products[Rodríguez-Gómez, et al, 2025].

EMA regulatory initiatives have focused on developing specialized expertise and collaborative frameworks for nanomedicine evaluation. The European Medicines Agency has established the PRIME (PRI orityMEdicines) scheme to enhance support for nanomedicine development targeting unmet medical needs. This voluntary scheme provides enhanced interaction and early dialogue with developers to optimize development strategies and accelerate evaluation timelines.[NNOTE-01].

The qualification of novel methodologies has become a critical component of nanomedicine regulation, with both agencies providing guidance on physiologically based pharmacokinetic (PBPK) modeling, in silico methods, and in vitro validation procedures. These methodologies enable more efficient evaluation of nanomedicine products while reducing reliance on traditional animal studies and clinical investigations.ema.europa

Harmonization efforts between FDA and EMA continue to evolve, with international collaborative initiatives aimed at developing consistent regulatory approaches for nanomedicine evaluation. These efforts address the global nature of nanomedicine development and the need for standardized evaluation criteria across regulatory jurisdictions.ema.europa

Personalized Medicine Regulatory Considerations

The integration of personalized medicine approaches with nanomedicine creates additional regulatory complexities that require specialized consideration. Regulatory agencies must address the unique challenges posed by individualized treatments that rely on patient-specific biomarkers, genetic profiles, and companion diagnostics [Dharani and Kamraj, 2024].

Biomarker validation represents a critical regulatory requirement for personalized nanomedicines. Regulatory agencies emphasize rigorous validation of biomarkers used to identify patients likely to benefit from specific nanomedicine treatments. The accuracy and clinical utility of these biomarkers directly impact therapeutic decision-making and patient outcomes, requiring comprehensive validation studies demonstrating analytical and clinical validity [Padamati, and Rangnath, 2023].

Companion diagnostic development has become integral to personalized nanomedicine approval pathways. These diagnostic tests identify patients most likely to benefit from particular nanomedicine therapies and are often co-developed with therapeutic products. Regulatory approval requires demonstration of the diagnostic test's ability to accurately predict treatment response and support clinical decision-making [Dharani and Kamraj, 2024]. The design of clinical trials for personalized nanomedicines requires careful consideration of patient stratification strategies based on biomarker profiles. Regulatory agencies provide guidance on appropriate trial designs that demonstrate effectiveness in specific patient populations while accounting for the heterogeneity inherent in personalized medicine approaches. Adaptive trial designs and innovative statistical approaches may be necessary to address the unique challenges of personalized nanomedicine evaluation.

Post-market surveillance assumes particular importance for personalized nanomedicines, given the potential for differential responses across patient populations. Regulatory frameworks emphasize ongoing monitoring of safety and efficacy in real-world clinical settings, with requirements for post-market studies that assess long-term outcomes and identify potential subpopulation-specific effects [Padamati, and Rangnath, 2023].

Clinical Trial Design Challenges

The design of clinical trials for nanomedicines presents unique methodological challenges that differ substantially from traditional pharmaceutical development. The complexity of nanomedicine products, their novel mechanisms of action, and sophisticated delivery systems require innovative trial design approaches that address both efficacy and safety considerations [Gawne, et al, 2023].

Preclinical-clinical translation represents a major challenge in nanomedicine development, with poor correlation between animal model results and clinical outcomes. Most nanomedicine candidates demonstrate promising preclinical efficacy but fail in clinical trials due to inadequate understanding of nanoparticle behavior in humans. The differences between animal models and human physiology, particularly in terms of immune system responses and pharmacokinetics, limit the predictive value of preclinical studies [Zheng, et al, 2021].

Endpoint selection and measurement in nanomedicine trials requires careful consideration of both traditional efficacy measures and nanomedicine-specific parameters. Clinical trials must evaluate not only therapeutic outcomes but also nanoparticle biodistribution, accumulation in target tissues, and clearance mechanisms. Advanced imaging techniques and bioanalytical methods may be necessary to assess these unique aspects of nanomedicine performance [Rehan, et al, 2024].

The complexity of nanomedicine formulations creates challenges in clinical trial standardization and reproducibility. Chemistry, Manufacturing, and Controls (CMC) requirements for nanomedicines exceed those of traditional drugs due to the intricate synthesis processes and multiple critical quality attributes. Batch-to-batch variability in nanomedicine products can significantly impact clinical trial results, requiring robust quality control measures and detailed characterization protocols [Zheng, et al, 2021].

Patient population definition for nanomedicine trials may require novel approaches that consider both disease characteristics and patient-specific factors affecting nanoparticle behavior. Factors such as immune system status, organ function, and genetic variants affecting nanoparticle metabolism may influence treatment outcomes, necessitating careful patient stratification strategies and potentially smaller, more defined patient populations [Gawne, et al, 2023].

6.4 Economic and Healthcare System Integration

Cost-Effectiveness Analysis

The economic evaluation of nanomedicines has become increasingly sophisticated, with comprehensive cost-effectiveness analyses revealing both opportunities and challenges in healthcare system integration. The global nanomedicine market, valued at USD 294.04 billion in 2024 and projected to reach USD 779.19 billion by 2033, represents substantial economic potential alongside significant cost considerations [Zhang, et al, 2024].

Development and production costs for nanomedicines significantly exceed those of traditional pharmaceuticals, with increases of approximately 15% compared to conventional drugs. Clinical trial costs alone range from \$101.2 billion to \$174.4 billion, with the intricate nature of nanomedicines contributing to cost escalation. These elevated expenses pose particular challenges for smaller research institutions and companies, creating barriers to innovation and market entry [Zhang, et al, 2024].

The cost-effectiveness of nanomedicine interventions varies substantially across therapeutic applications, with cancer treatment and rare diseases showing the most favorable economic profiles. The nanomedicine market's projected compound annual growth rate of 11.57% from 2023 to 2030 reflects increasing recognition of therapeutic value, though cost-effectiveness remains highly dependent on specific clinical applications and patient populations.grandviewresearch+1

Strategies for cost reduction include simplifying manufacturing processes, enhancing collaborative efforts between academic institutions and industry, and implementing clear regulatory guidelines to reduce compliance costs. Artificial intelligence and machine learning applications show promise in

expediting drug candidate identification and reducing clinical trial scope, while advanced manufacturing technologies such as continuous processing can increase production efficiency and reduce waste [Zhang, et al, 2024].

Value-based pricing models are emerging as important mechanisms for nanomedicine reimbursement, focusing on demonstrated clinical outcomes rather than traditional cost-plus approaches. These models attempt to align nanomedicine costs with therapeutic value, though implementation challenges remain significant across different healthcare systems and payer organizations [Mahajan, and Powell, 2025].

Healthcare Infrastructure Requirements

The integration of nanomedicines into healthcare systems requires substantial infrastructure adaptations to support safe and effective clinical implementation. Healthcare facilities must develop specialized capabilities for nanomedicine storage, handling, administration, and monitoring that differ significantly from traditional pharmaceutical requirements [Malik, et al, 2023].

Specialized storage and handling facilities are necessary for many nanomedicine products due to their unique stability requirements and potential safety considerations. Temperature-controlled environments, specialized packaging systems, and contamination prevention measures represent significant infrastructure investments for healthcare institutions. These requirements are particularly challenging for smaller healthcare facilities with limited resources [Zhang, et al, 2024].

Healthcare workforce training represents a critical infrastructure requirement for nanomedicine implementation. Medical professionals, pharmacists, and support staff require specialized education on nanomedicine properties, administration techniques, monitoring requirements, and adverse event management. This training represents ongoing costs and resource allocation challenges for healthcare systems [Malik, et al, 2023].

Advanced monitoring and diagnostic capabilities are often necessary for optimal nanomedicine utilization. Lab-on-chip technologies, nanotechnology-enhanced diagnostic tools, and real-time monitoring systems require significant capital investments and technical expertise. These technologies enable personalized treatment approaches and improved patient outcomes but demand substantial healthcare system modifications [Malik, et al, 2023].

The distribution and supply chain infrastructure for nanomedicines requires specialized logistics capabilities due to stability requirements, cold chain management, and quality assurance protocols. Healthcare systems must invest in appropriate storage and transportation systems while ensuring product integrity throughout the distribution process [Zhang, et al, 2024].

Reimbursement Challenges

Reimbursement frameworks for nanomedicines face substantial challenges due to the complexity of these therapeutic approaches and limitations of existing payment systems. Current healthcare payment structures, designed around discrete medical services and procedures, struggle to capture the multifaceted value propositions offered by nanomedicine technologies.[NMMR-01].

The integration of generalist nanomedicine systems presents particular reimbursement challenges, as these technologies may perform multiple clinical tasks simultaneously rather than single, discrete functions. Traditional billing codes and payment structures may inadequately reflect the comprehensive capabilities and efficiency gains provided by advanced nanomedicine platforms, requiring innovative payment model development [Mahajan, and Powell, 2025].

Value-based reimbursement approaches are emerging as potential solutions for nanomedicine coverage, though implementation remains complex. These models attempt to tie reimbursement to demonstrated clinical outcomes and cost-effectiveness rather than traditional fee-for-service approaches. However, establishing appropriate outcome measures and payment levels requires extensive economic analysis and stakeholder collaboration [Mahajan, and Powell,2025].

Equity and access considerations represent critical challenges in nanomedicine reimbursement policy development. High development costs and complex manufacturing requirements can result in expensive nanomedicine products that may create access disparities. Reimbursement frameworks must balance innovation incentives with equitable patient access, potentially requiring targeted subsidy programs or risk-sharing arrangements [Mahajan, and Powell,2025].

International variation in reimbursement approaches creates additional challenges for global nanomedicine development and commercialization. Different healthcare systems employ varying reimbursement methodologies, coverage criteria, and economic evaluation requirements, complicating global product development strategies and market access planning [Mahajan, and Powell, 2025].

The regulatory approval and reimbursement coordination remains a significant challenge, as regulatory approval does not guarantee reimbursement coverage. Payer organizations conduct independent economic evaluations that may reach different conclusions than regulatory agencies regarding product value and appropriate utilization, creating potential barriers to patient access even for approved nanomedicine products [NMMR-01].

The successful clinical translation of nanomedicines requires addressing these multifaceted challenges through coordinated efforts among researchers, regulatory agencies, healthcare systems, and payer organizations. While substantial obstacles remain, the potential benefits of nanomedicine technologies continue to drive innovation and investment in overcoming these barriers to clinical implementation.

7. Current Clinical Trials and Commercial Developments

The clinical translation of 2D nanomaterials represents a pivotal moment in nanomedicine, with groundbreaking human trials, emerging commercial products, and valuable lessons learned from both successes and failures. This landscape is characterized by accelerating innovation, substantial market growth, and increasing regulatory clarity that collectively position 2D materials as transformative technologies in healthcare delivery.

7.1 Ongoing Clinical Studies

Phase I/II Trials with 2D Nanomaterials

The first-ever human clinical trial involving 2D nanomaterials achieved a historic milestone in 2024, when researchers from the Universities of Edinburgh and Manchester conducted a controlled inhalation study of graphene oxide in healthy volunteers. This groundbreaking double-blind randomized controlled study involved 14 participants who inhaled carefully controlled doses of thin, ultra-pure graphene oxide nanosheets for two hours while cycling in a purpose-designed mobile exposure chamber [FHTRIAL].

The study demonstrated that very pure forms of graphene oxide with specific size distribution and surface characteristics showed no short-term adverse effects on lung function, blood pressure, blood clotting, or inflammation markers. This represents the first controlled study involving healthy people to demonstrate that graphene-based materials can be developed safely, providing crucial safety data for future clinical applications. The research laid the foundation for subsequent human studies investigating larger populations and different graphene formulations [Andrews, et al, 2024].

Clinical investigations of graphene electrodes are advancing rapidly, with ongoing trials examining brain mapping applications during surgical procedures. The NCT06368310 clinical trial is evaluating the safety of graphene-based electrodes for brain mapping during tumor resection surgeries, marking a significant step toward clinical implementation of 2D materials in neurosurgical applications [CLINTRIAL].

Recent developments in 2D materials for cancer therapeutics have progressed to clinical evaluation stages, with multiple Phase I/II trials investigating graphene-based drug delivery systems and photothermal therapy applications. Studies demonstrate that graphene nanosheets can achieve 2-3 orders of magnitude improvement in chemotherapeutic effectiveness against HCT-116 colorectal cancer cells compared to conventional treatments [Uzdrowska, et al. 2024]

Rare earth-based 2D materials have entered preclinical stages with promising biomedical applications in imaging, therapy, and monitoring. These materials leverage unique magnetic, optical, and catalytic properties enhanced by the 2D structure, offering boundless possibilities for advanced diagnostic and therapeutic platforms [Zhu, et al, 2025].

Patient Stratification Strategies

The development of biomarker-guided patient stratification has become essential for optimizing 2D nanomaterial clinical trials. Modern cancer nanomedicine trials increasingly employ sophisticated patient selection criteria based on genetic profiles, tumor characteristics, and biomarker expression patterns to identify individuals most likely to benefit from specific 2D material-based therapies [van der Meel, et al, 2019].

Circulating tumor DNA (ctDNA) analysis has emerged as a powerful tool for patient stratification in nanomedicine trials. The GOZILA study demonstrated that patients treated with biomarker-matched targeted therapy based on ctDNA profiling showed significantly improved overall survival compared to those receiving unmatched therapy, with a hazard ratio of 0.54. This approach enables precise identification of patients suitable for specific 2D nanomaterial interventions [Nakamura, et al, 2025].

Adaptive trial designs are increasingly employed in 2D nanomaterial studies to accommodate the complexity of patient stratification. Examples include multi-arm trials like ATLANTIS, which explores maintenance targeted therapy after chemotherapy with treatment randomization based on biomarker profiles, and PRIMUS001, which assesses efficacy in both biomarker-positive and unselected patient populations[Antoniou, et al, 2019].

The integration of companion diagnostics with 2D nanomaterial therapies has become a critical component of clinical trial design. These diagnostic tests identify patients most likely to benefit from particular treatments and are often co-developed with therapeutic products to ensure optimal patient selection and treatment outcomes[Antoniou, et al, 2019].

Molecular profiling approaches are being refined to address the unique properties of 2D nanomaterials and their interactions with biological systems. Patient stratification strategies now consider factors such as immune system status, organ function, and genetic variants affecting nanoparticle metabolism, which may influence treatment outcomes and require careful consideration in trial design[Antoniou, et al, 2019].

Preliminary Efficacy Data

Graphene-based cancer therapeutics have demonstrated promising preliminary efficacy in early-phase clinical studies. Surface-tailored graphene nanosheets targeting PI3K/Akt signaling pathways have shown significant anticancer activity against breast cancer cell lines, with dose-dependent reduction in colony-forming ability confirmed through clonogenic assays [Durgadevi, and Kumar, 2025].

Photothermal therapy applications using 2D materials have achieved remarkable therapeutic outcomes in preclinical and early clinical studies. Graphene-based composites demonstrate highly effective light-to-heat conversion capabilities, with studies showing significant reduction in tumor cell viability from 40% to 5% after 48 hours of treatment with optimized dosing regimens. [Uzdrowska, et al, 2024].

Aqueous graphene formulations have demonstrated selective cytotoxicity against cancer cells while maintaining low toxicity to normal tissues. Recent studies show that few-layer graphene flakes produced through benign ultrasonic methods achieved approximately 30% cancer cell death at low concentrations (0.12-0.36 μg/ml) comparable to conventional chemotherapy agents like Reversine [Kaur, et al, 2025].

Drug delivery applications utilizing 2D materials have shown enhanced therapeutic efficacy across multiple disease models. Recent advances demonstrate that 2D material-based gene delivery nanosystems achieve high therapeutic efficiency for cancer treatment, with improved transfection efficiency compared to conventional non-viral vectors such as liposomes [Zhang, et al, 2024].

Neurological applications of 2D nanomaterials have produced encouraging preliminary results in treating neurodegenerative diseases. Studies demonstrate that graphene oxide enhances autophagy and increases ubiquitination of mutant huntingtin protein, showing potential for Huntington's disease treatment, while MoS₂ quantum dots mitigate amyloid beta aggregate-mediated neurotoxicity in Alzheimer's disease models [Li, et al, 2023].

7.2 Commercial Products and Market Analysis

FDA-Approved 2D Material-Based Devices

While no purely 2D material-based medical devices have received full FDA approval as standalone products, several 2D material-enhanced devices have gained regulatory clearance through established pathways. The FDA's approach to 2D materials follows existing medical device regulations, with enhanced scrutiny for novel nanomaterial applications requiring comprehensive biocompatibility and safety documentation [FDA].

Graphene-enhanced medical devices are progressing through FDA evaluation pathways, with particular focus on neural interface applications and diagnostic sensors. The AI-Enabled Medical Device List maintained by the FDA includes several devices incorporating advanced nanomaterials, though specific 2D material classifications remain under development [FDA].

Current regulatory frameworks address 2D materials under existing guidance for nanomedicines and novel medical devices. The FDA's Medical Device Material Safety Summaries provide comprehensive evaluation criteria for materials commonly used in implantable devices, establishing foundation requirements that 2D materials must meet for clinical approval [FDA].

International regulatory coordination is advancing through collaborative frameworks between FDA and EMA for 2D material evaluation. Recent approvals of nanomaterial-based devices provide precedent and regulatory pathways that 2D material developers can leverage for expedited approval processes [FDA].

Market Projections and Trends

The global nanomedicine market has experienced remarkable growth, valued at USD 241.82 billion in 2024 and projected to reach USD 570.98 billion by 2032, exhibiting a compound annual growth rate of 11.7%. This expansion reflects increasing adoption of advanced nanomaterials, including 2D materials, across therapeutic and diagnostic applications [NMMS].

Graphene-based biomedical applications represent a rapidly expanding market segment, with the global graphene market estimated at USD 195.7 million in 2023 and projected to reach USD 1,609.3 million by 2030, growing at a CAGR of 35.1%. The electronics and energy storage sectors drive significant demand, with biomedical applications emerging as a high-growth segment [GMS].

Regional market dynamics show North America dominating nanomedicine markets with 45.88% market share in 2024, followed by significant growth in Asia-Pacific regions. India's graphene market, valued at USD 9.86 million in 2024, is projected to reach USD 138.35 million by 2033 with a CAGR of 31.58%, reflecting emerging market potential [INDMS].

2D carbon material markets specifically show strong growth trajectories, with the 2D carbon material graphene market valued at USD 2.67 billion in 2024 and projected to reach USD 6.98 billion by 2033, exhibiting a CAGR of 11.4%. This growth is driven by expanding applications in electronics, energy storage, and emerging biomedical sectors [VMR].

Investment trends indicate substantial funding for nanomedicine research and development, with leading companies focusing on strategic partnerships, product launches, and clinical trial advancement. The global nanomedicine market is projected to grow from USD 218.25 billion in 2024 to USD 767.15 billion by 2035 at a CAGR of 12.11% [NMCOM].

Key Industry Players

NanoXplore Inc. (TSXV:GRA) has emerged as a leading graphene material producer, with market capitalization of C\$411.17 million and reported Q2 2025 revenues of C\$33.12 million, representing 14% growth year-over-year. The company's GrapheneBlack graphene powder and SiliconGraphene battery anode materials position it as a key supplier for automotive and energy storage applications [INVEST].

Black Swan Graphene (TSXV:SWAN) represents an emerging powerhouse in bulk graphene production, with market capitalization of C\$33.19 million and strategic partnership with UK-based Thomas Swan & Co., which holds 15% interest and provides patent portfolio and intellectual property. The company launched GraphCore 01 family products in 2024 and secured C\$6 million equity financing in February 2025 for capacity expansion [GSTOCK].

Haydale Graphene Industries (LSE:HAYD) focuses on commercializing proprietary heating ink-based technology and integrating graphene into next-generation industrial applications. The company maintains strategic partnerships with University of Manchester's Graphene Engineering Innovation Centre and has secured commercial contracts for heating systems and gas network applications [INVEST].

First Graphene Ltd. (ASX:FGR) operates as a vertically integrated mine-to-graphene producer, developing PureGRAPH® product lines for industrial additives and participating in consortiums developing graphene-enhanced composite tanks for liquid hydrogen storage. The company recently secured 5-year distribution agreements and funding for scaling Kainos technology [GSTOCK].

Zentek Ltd. (TSXV:ZEN) focuses on graphene applications in antimicrobial and air filtration technologies, targeting healthcare and cleantech markets. The company's ZenGuard coating technology addresses infection control needs, though commercial uptake remains in early stages due to regulatory requirements [GSTOCK].

Archer Materials (ASX:AXE) pioneers graphene transistor biochip development for point-of-care diagnostics, partnering with UK's Paragraf to accelerate biochip development for kidney disease monitoring. The company's graphene field-effect transistor arrays enable detection of biomolecules at very low concentrations [GSTOCK].

7.3 Success Stories and Lessons Learned

Notable Success Stories

The University of Manchester's first human graphene trial represents a landmark achievement in 2D nanomaterial clinical translation. This carefully controlled study demonstrated that ultra-pure graphene oxide can be developed safely for human exposure, opening pathways for future clinical

applications. The research team's decade-long preparation, including extensive preclinical investigations and international collaboration, exemplifies best practices for translating laboratory innovations to clinical reality [FHTRIAL].

Iron oxide nanoparticles (IONPs) provide valuable precedent as successful nanomaterials in clinical translation, with numerous FDA-approved products for therapeutics and imaging applications. These successes demonstrate that nanomaterials can achieve regulatory approval and commercial success when proper safety profiles are established and clinical benefits are clearly demonstrated [Hassan, et al, 2017].

Abraxane's commercial success in cancer nanomedicine illustrates effective clinical translation strategies that 2D materials can emulate. Unlike failed nanomedicine candidates CRLX101 and BIND-014, Abraxane succeeded by focusing on maximum tolerated dose optimization and demonstrating clear clinical advantages over existing therapies [Mukherjee, et al, 2022].

Graphene electrode applications in neurosurgery represent emerging success stories, with ongoing clinical trials demonstrating safety and efficacy for brain mapping during tumor resection procedures. These applications leverage graphene's superior electrical properties and biocompatibility to improve surgical outcomes and patient safety [Gravagnuolo, et al, 2024].

Critical Lessons from Failures

The high attrition rate in nanomedicine clinical trials provides crucial lessons for 2D material development. Analysis reveals that 90% of clinical drug development fails, with 40-50% due to lack of efficacy, 30% from unmanageable toxicity, and 10-15% from poor drug-like properties. For nanomedicines specifically, Phase II trial success rates drop to approximately 48%, with Phase III success rates falling to merely 14% [Sun, et al. 2022].

Translation challenges from preclinical to clinical settings highlight critical gaps in current development approaches. The biological discrepancy between animal models and human physiology often leads to failure of promising preclinical candidates in clinical trials. This emphasizes the need for more predictive preclinical models and better understanding of human-specific responses to 2D nanomaterials [Mukherjee, et al, 2022].

Target validation challenges represent a fundamental issue affecting nanomedicine success rates. Many drug candidates fail because their intended molecular targets differ from their actual biological targets, or because target validation in animal models doesn't translate to human disease contexts. For 2D materials, this emphasizes the importance of understanding their multifaceted biological interactions beyond single-target approaches [Sun, et al. 2022]

Manufacturing and quality control issues have caused significant setbacks in nanomedicine development. Batch-to-batch variability and scaling challenges that become apparent only during clinical development highlight the need for robust manufacturing processes established early in development programs [Mukherjee, et al, 2022].

Best Practices for Clinical Translation

Rigorous safety assessment protocols emerge as essential for successful 2D material translation. The Manchester graphene trial's success relied on extensive preclinical safety studies, careful material characterization, and conservative exposure parameters. This approach should be standard for all 2D material clinical development programs [Andrews, et al, 2024].

Strategic patient stratification has proven critical for nanomedicine success. Biomarker-guided approaches that identify optimal patient populations early in development can significantly improve clinical trial outcomes and accelerate regulatory approval pathways [van der Meel, et al, 2024].

Integrated development approaches combining materials science, clinical medicine, and regulatory expertise from early development stages facilitate more successful translation. Companies achieving commercial success typically invest in interdisciplinary teams and regulatory guidance early in development processes [Hassan, et al, 2017].

Focus on unmet medical needs represents a common theme among successful nanomedicine programs. Products addressing clear clinical needs with superior efficacy or safety profiles achieve higher success rates than those offering incremental improvements over existing therapies [Hassan, et al, 2017].

Collaborative research models involving academic institutions, industry partners, and regulatory agencies have proven effective for advancing 2D material applications. The University of Manchester's partnerships with industry and international research institutes exemplify successful collaborative approaches that accelerate clinical translation while maintaining scientific rigor [FHTRIAL].

The clinical translation of 2D nanomaterials is rapidly advancing, with first human trials demonstrating safety, emerging commercial applications, and valuable lessons learned from both successes and failures. While challenges remain significant, the combination of increasing regulatory clarity, substantial market opportunities, and growing clinical evidence suggests that 2D materials are poised to make substantial contributions to future healthcare delivery. Success will depend on continued emphasis on safety, strategic patient selection, robust manufacturing processes, and collaborative development approaches that leverage the unique properties of these remarkable materials.

8. Future Perspectives and Research Directions

The future landscape of 2D nanomaterials in biomedical applications presents unprecedented opportunities for transformative healthcare delivery through emerging technologies, precision medicine integration, and systematic addressing of current limitations. This comprehensive vision encompasses revolutionary advances in material science, personalized therapeutic approaches, and global healthcare transformation that will fundamentally reshape medical practice by 2030-2040.

8.1 Emerging Technologies and Innovations

Next-Generation 2D Materials

The development of next-generation 2D materials is rapidly expanding beyond traditional graphene applications to encompass a diverse palette of ultrathin materials with unique properties tailored for specific biomedical applications. Transition metal dichalcogenides (TMDs), MXenes, layered double hydroxides (LDHs), metal-organic frameworks (MOFs), and covalent organic frameworks (COFs) represent the forefront of emerging 2D material platforms that offer unprecedented opportunities for therapeutic and diagnostic applications [Murali, et al, 2021].

Rare earth-based 2D materials have emerged as particularly promising candidates for advanced biomedical applications, leveraging unique magnetic, optical, and catalytic properties enhanced by their 2D structure. These materials offer boundless possibilities for advanced diagnostic and therapeutic platforms, with applications ranging from high-resolution medical imaging to targeted drug delivery systems. Recent developments demonstrate that rare earth elements integrated into 2D frameworks can provide superior contrast enhancement for magnetic resonance imaging while maintaining excellent biocompatibility profiles [Zhu, et al, 2025].

MXenes represent a revolutionary class of 2D materials with exotic and superior properties, consisting of transition metal carbides, carbonitriles, and nitrides that have captured global research attention. The SAFARI project exemplifies cutting-edge development approaches, focusing on safety and sustainability aspects while developing hybrid formulations of MXene-Graphene materials. These hybrid systems exhibit excellent electrical, electrochemical, and electromagnetic properties, creating safer 2D material configurations for industrial applications due to their larger size (\sim 1-2 μ m), which reduces potential health risks [ARTICLE-G].

Sustainable production routes for next-generation 2D materials are advancing through innovative approaches that combine Spark Plasma Sintering (SPS) and High-Energy Ball Milling (HEBM) pilot lines. The scale-up production of MXenes phases is being achieved through High Frequency Acoustic Emission (HFAE) processes, representing fast and environmentally friendly manufacturing without toxic acid usage. These developments address critical sustainability concerns while enabling commercial-scale production of high-quality 2D materials[ARTICLE-G].

Hybrid Nanoplatforms

The development of hybrid nanoplatforms combining 2D materials with complementary nanoscale components represents a paradigm shift toward multifunctional therapeutic systems. These sophisticated platforms integrate the unique properties of different nanomaterials to create synergistic effects that exceed the capabilities of individual components, enabling simultaneous therapeutic delivery, imaging, and monitoring functions [Davis, et al, 2021].

Gold nanoparticle-graphene hybrid systems exemplify the potential of hybrid nanoplatforms for multimodal cancer therapy and imaging. These platforms combine the photothermal therapy effects of both gold nanoparticles and graphene derivatives, achieving synergistic photothermal effects while providing enhanced imaging capabilities through fluorescence, Raman, and thermal imaging techniques. The high surface areas of graphene oxide and reduced graphene oxide enable efficient drug and fluorophore loading, creating comprehensive theranostic platforms that can simultaneously treat and visualize cancer progression [Holca, et al, 2025].

Stimuli-responsive hybrid nanoplatforms are being engineered to respond to both endogenous factors (pH, redox conditions, enzyme activity) and exogenous stimuli (thermal effects, magnetic fields, light, ultrasound). These smart platforms enable precise spatial, temporal, and dosage control over therapeutic release, minimizing toxicity to normal tissues while maximizing therapeutic efficacy at target sites. The integration of multiple responsive mechanisms creates sophisticated "on-off" functionalities that can be precisely controlled based on specific physiological conditions [Chen, et al, 2024]. Bio-integrated hybrid platforms combine 2D materials with biological components such as proteins, DNA, and cellular membranes to create biomimetic systems with enhanced biocompatibility and targeting specificity. These biohybrid platforms leverage natural biological recognition mechanisms while incorporating the superior physical and chemical properties of 2D materials, creating systems that can seamlessly interface with biological environments while delivering therapeutic payloads with unprecedented precision [Gravagnuolo, et al, 2024].

Smart Responsive Systems

The evolution toward smart responsive systems represents the convergence of 2D materials with advanced sensing, computing, and actuation capabilities. These intelligent platforms can autonomously monitor physiological conditions, make therapeutic decisions, and adjust treatment parameters in real-time, representing the future of personalized medicine delivery systems [Zhao, et al, 2024].

Stimuli-responsive nanomaterials equipped with triggered modules enable sophisticated drug delivery systems that can respond to specific biological conditions. These systems utilize endogenous stimuli including pH variations, redox gradients, and enzymatic activity within pathological microenvironments, as well as externally applied stimuli such as thermal effects, magnetic fields, light irradiation, and ultrasound activation. The ultimate objective is precise drug release control while minimizing toxicity to healthy tissues [Chen, et al, 2024].

Piezoelectric and flexoelectric 2D materials enable the development of self-powered biomedical devices that can harvest energy from biological movements and respond to mechanical stimuli. TMDs demonstrate exceptional piezoelectric and flexoelectric effects that create new opportunities in biosensing and tissue regeneration applications. These materials can respond to voltage changes associated with strain gradients, making them particularly valuable for bone regeneration applications where physiological repair processes involve piezoelectric and flexoelectric voltage changes [Wu, et al, 2025].

Bio-integrated soft electronics utilizing 2D materials enable non-invasive in situ monitoring of physiological variables while providing large surface areas for cellular interactions. These systems create unique bridges between nano- and micro-scale biological systems, offering real-time monitoring capabilities combined with therapeutic intervention potential. The development of flexible, wearable, and implantable smart systems will enable continuous health monitoring and autonomous therapeutic adjustments based on individual patient needs [Bolotsky, et al, 2019]

8.2 Integration with Precision Medicine Initiatives

Genomics and Proteomics Integration

The integration of genomics and proteomics data with 2D nanomaterial platforms represents a fundamental shift toward molecularly-informed therapeutic design. Advanced bioinformatics approaches enable the combination of genetic information with protein expression profiles to create comprehensive molecular blueprints that guide nanomaterial design and therapeutic targeting strategies [Elrashedy, et al, 2025].

Multi-omics data integration utilizing ratio-based quantitative profiling approaches enables reproducible and comparable data suitable for integration across batches, laboratories, platforms, and omics types. The Quartet Project provides multi-omics ground truth and best practices for quality control and data integration, establishing reference materials that include DNA, RNA, protein, and metabolite profiles derived from family quartets. This systematic approach enables objective evaluation of horizontal integration within single omics types and vertical integration across multiple omics platforms [Zheng, et al, 2024].

Proteomics analysis pipelines such as ProtPipe enable multifunctional data analysis for high-throughput proteomics datasets, supporting data quality control, imputation, sample filtering, and normalization. These comprehensive platforms facilitate pathway enrichment analysis, protein-protein interaction studies, and MHC-peptide binding affinity prediction, providing essential tools for translating proteomic insights into nanomaterial design parameters and therapeutic targeting strategies [Li, et al, 2024].

Genomics-first approaches layered with other omics data offer practical models for adopting integrated multi-omics approaches in personalized healthcare. The genotype-first approach or reverse phenotyping enables identification of new genotype-phenotype associations, enhanced disease subclassification through widened phenotypic spectra, and improved understanding of functional mechanisms underlying genetic variations. This approach provides critical guidance for designing 2D nanomaterial systems that can respond to specific genetic profiles and molecular signatures [Mani, et al, 2025].

Pharmacogenomics Applications

The convergence of pharmacogenomics with nanomedicine creates unprecedented opportunities for nano-enabled personalized therapeutics that account for individual genetic variations in drug metabolism and response. This integration enables the development of targeted delivery systems that can adjust drug release kinetics, targeting specificity, and therapeutic mechanisms based on patient-specific genetic profiles [Tiwari, et al, 2025].

Nano-enabled pharmacogenomics represents the integration of pharmacogenomic sciences with advanced nanotechnology to create highly sophisticated personalized medicine platforms. This approach addresses traditional pharmacogenomic challenges including drug resistance and non-specific distribution by utilizing nanoparticle delivery systems that can overcome biological barriers and provide targeted therapeutic delivery. The field encompasses gene editing tools, personalized drug formulations, and companion diagnostic systems that work synergistically to optimize therapeutic outcomes [Lee, 2025].

CRISPR-Cas9 integration with nanomedicine platforms enables precise genetic modifications targeted to specific disease pathways. Gold nanoparticles have been employed to deliver siRNA targeting dopamine transporters, effectively reducing drug-seeking behavior in preclinical addiction models. Exosome-based nanocarriers show potential for delivering neurotrophic factors such as BDNF and GDNF to promote neuroprotection and neuroregeneration in addiction-affected brain regions, opening pathways for personalized treatments tailored to individual genetic profiles. [Patne, et al, 2025].

Pharmacogenomic biomarker integration enables rational drug selection for nanomedicine applications, particularly in cancer treatment where molecular targets must be precisely matched with therapeutic interventions. The incorporation of biomarkers such as dopamine transporter density, opioid receptor expression, and inflammatory markers enables personalization of treatment regimens for improved clinical outcomes. This approach ensures that nanomedicine platforms are designed to interact optimally with individual patient biology based on genetic predisposition and molecular expression patterns [Ahmad, and Mohammad, 2021].

Multi-Omics Approaches

The implementation of comprehensive multi-omics strategies enables holistic understanding of disease mechanisms and treatment responses by integrating genomics, transcriptomics, proteomics, metabolomics, and epigenomics data. This approach provides unprecedented insights into complex biological processes that guide the design of sophisticated 2D nanomaterial platforms capable of addressing multiple therapeutic targets simultaneously [Ikwelle, et al, 2025].

Network medicine approaches utilizing protein-protein interactomes and biological networks enable systematic integration of multi-omics data at different biological levels to enhance precision medicine applications. These frameworks provide unbiased methodologies for identifying disease pathways and deciphering relationships among drugs, their targets, and diseases. The integration spans patient-level to single-cell-level analysis, enabling comprehensive understanding of therapeutic mechanisms and optimization strategies [Wang, et al, 2023].

Artificial intelligence-enhanced multi-omics integration leverages machine learning and advanced computational approaches to identify patterns and relationships across diverse biological datasets. These AI-driven approaches enable the discovery of novel biomarkers, therapeutic targets, and personalized treatment strategies that would be impossible to identify through traditional single-omics approaches. The integration of AI with multi-omics data creates opportunities for predictive modeling that can guide nanomaterial design and therapeutic optimization [Molla and Bitew, 2024].

Translational multi-omics platforms bridge the gap between molecular understanding and clinical implementation by providing frameworks for translating complex biological insights into actionable therapeutic strategies. These platforms enable the identification of patient-specific molecular signatures that can guide the selection and customization of 2D nanomaterial-based therapeutic interventions, ensuring optimal matching between individual patient biology and nanomaterial properties.

8.3 Addressing Current Limitations

Standardization Needs

The establishment of comprehensive standardization protocols represents a critical requirement for advancing 2D nanomaterials from research applications to clinical implementation. Current limitations in standardized synthetic protocols, characterization methods, and safety assessment procedures create significant barriers to reproducible research and regulatory approval processes [Dai, 2024].

Standardized synthetic protocols are essential for ensuring consistent material properties and structural control across different production facilities and research laboratories. The development of scalable manufacturing approaches including chemical vapor deposition (CVD), exfoliation methods, and wet-chemical synthesis requires comprehensive standardization to achieve uniform particle size distribution, surface functionalization, and physicochemical properties. Current challenges include variations in reaction parameters, including temperature, time, precursor concentrations, and solvents, which significantly affect material quality and reproducibility [Zhang, 2015].

Comprehensive characterization standards must address the unique properties of 2D materials including lateral size, thickness, surface functionalization, crystal structure, and aggregation behavior. Complete characterization of these properties is essential for accurately assessing hazard potential and biological interactions. The development of standardized analytical methods utilizing advanced techniques such as atomic force microscopy, transmission electron microscopy, X-ray photoelectron spectroscopy, and Raman spectroscopy is critical for ensuring consistent material evaluation across research institutions and regulatory agencies [Guiney, et al, 2018].

Regulatory harmonization efforts must address the unique challenges posed by 2D nanomaterials while building upon existing frameworks for nanomedicine evaluation. The development of standardized protocols for toxicity testing and environmental safety assessments requires multidisciplinary collaboration among materials scientists, biomedical researchers, regulatory experts, and clinical practitioners. These standardized approaches will ensure that the benefits of 2D materials can be harnessed without compromising health and environmental integrity [Dai, 2024].

Long-Term Safety Assessment

The comprehensive evaluation of long-term safety represents one of the most critical challenges facing 2D nanomaterial translation to clinical applications. Current understanding of biodistribution, accumulation, degradation, and elimination pathways remains limited, creating significant uncertainties regarding chronic exposure effects and long-term biocompatibility [ARTICLE-01].

Degradation pathway analysis has become increasingly sophisticated, revealing that 2D materials can undergo transformation both in environmental conditions and within biological systems. Environmental degradation occurs through exposure to air, sunlight, water, and biological decomposers including bacteria, fungi, and insects. Within the human body, degradation products may affect cells differently than intact 2D materials, and non-degraded or slowly degrading materials may accumulate in vital organs, potentially impairing their function over time.rivm

Systematic toxicological evaluation requires comprehensive assessment across multiple biological levels, from cellular to organismal systems. In vitro studies must evaluate cytotoxicity, genotoxicity, and cellular responses, while in vivo studies using animal models assess inflammatory and immune responses, bioaccumulation patterns, and chronic effects. Environmental impact assessments must investigate release, distribution, and degradation in natural environments, understanding interactions with soil, water, and air systems [Guiney, et al, 2018].

Biocompatibility assessment frameworks must address the unique challenges posed by 2D materials' high surface-to-volume ratios, quantum effects, and ability to cross biological membranes. The development of predictive models based on structure-activity relationships will enable more efficient safety evaluation while reducing reliance on extensive animal testing. Long-term monitoring studies in clinical applications will provide essential real-world safety data to validate preclinical assessments and guide future material design strategies [Guiney, et al, 2018].

Scale-Up Challenges

The translation of 2D nanomaterials from laboratory to commercial production faces substantial technical, economic, and quality control challenges that must be systematically addressed to enable widespread clinical implementation. Current manufacturing approaches often fail to maintain precise control over material properties when scaled to industrial production levels.

Manufacturing scalability issues arise from fundamental differences between laboratory-scale synthesis and industrial production requirements. Top-down approaches involving mechanical size reduction require significant capital investment and energy consumption while struggling to maintain uniform properties. Bottom-up approaches building materials molecule by molecule offer greater control but face challenges in maintaining uniformity and yield at larger scales. The cost implications are substantial, with production costs for nanomaterials often orders of magnitude higher than bulk materials [Hu and Dong, 2024].

Quality control standardization becomes increasingly complex at commercial scales, requiring sophisticated analytical methods and real-time monitoring systems. The primary challenges include achieving economical and uniform production of defect-free 2D thin layers, synthesizing materials with specific size distributions and surface functionalization, and maintaining consistent physicochemical properties across production batches. Current manufacturing methods are either time-intensive or costly, lacking scalability for large-scale production [Hu and Dong, 2024].

Sustainable manufacturing approaches are being developed to address environmental and economic concerns associated with 2D material production. Continuous manufacturing processes operating under constant parameters can produce consistent quality products regardless of batch size, overcoming traditional limitations that become problematic at scale. Flow technology and microfluidic platforms offer particular advantages in process safety and quality control, though case-by-case evaluation remains necessary to determine optimal manufacturing approaches for specific material types and applications.

8.4 Vision for 2030-2040

Fully Personalized Treatment Paradigms

The vision for fully personalized treatment paradigms by 2030-2040 encompasses comprehensive integration of individual genetic profiles, molecular signatures, environmental factors, and lifestyle characteristics with advanced 2D nanomaterial therapeutic platforms. This transformation will enable unprecedented precision in therapeutic delivery, real-time treatment optimization, and predictive health management [Asma, 2025].

Super-personalized healthcare solutions will leverage cutting-edge genomic profiling, proteomics analysis, and digital health technologies to create individualized treatment strategies. The goal of precision medicine extends beyond genetic profile-based treatments to encompass comprehensive understanding of individual patient characteristics including molecular profiles, environmental exposures, and behavioral patterns. Advanced 2D nanomaterial platforms will be designed to respond dynamically to these individual characteristics, providing truly personalized therapeutic interventions [PNOTE-02].

Patient-specific nanomaterial design will enable the creation of therapeutic platforms tailored to individual molecular signatures and disease characteristics. Digital twin technologies will enable testing of therapeutic interventions in virtual patient models before clinical implementation, allowing optimization of nanomaterial properties, drug loading, release kinetics, and targeting strategies for each individual patient. This approach will maximize therapeutic efficacy while minimizing adverse effects through precise matching of nanomaterial characteristics with patient-specific biology [Goetz, et al. 2018].

Predictive personalized medicine will utilize advanced AI algorithms and multi-omics data integration to anticipate disease progression, treatment responses, and optimal intervention timing. By 2030, personalized medicine will be recognized as a medical specialty centered on individual patient characteristics, resulting in enhanced diagnostic, therapeutic, and preventative efficacy, increased economic value, and equitable access for all patients through five key aspects: prevention, diagnosis, treatment, monitoring, and prognosis [Sharma, et al, 2023].

AI-Driven Therapeutic Design

The integration of artificial intelligence with 2D nanomaterial design and therapeutic optimization will revolutionize healthcare delivery by 2030-2040, enabling autonomous therapeutic systems capable of real-time adaptation and optimization. This transformation will encompass predictive analytics, automated treatment protocols, and intelligent monitoring systems that collectively create a new paradigm of AI-empowered healthcare.

AI-enhanced material discovery is already accelerating the identification and optimization of novel 2D materials through computational screening and predictive modeling approaches. AI-driven computational materials science can predict properties of novel 2D materials with unprecedented accuracy, significantly reducing experimental trial requirements. Machine learning models can analyze complex experimental data, identify optimal synthesis parameters, and enhance yield and quality of materials like graphene and transition metal dichalcogenides, improving efficiency while reducing production costs.

Intelligent therapeutic platforms will combine AI-driven decision-making with 2D nanomaterial delivery systems to create autonomous therapeutic interventions. By 2030, AI and foundation model technologies will enable real-time analysis of vast pathological datasets, enabling more personalized treatment recommendations. AI-powered platforms will evaluate medication effectiveness, provide personalized prescriptions, and modulate therapeutic delivery based on continuous patient monitoring and predictive analytics [PNOTE].

Digital twin integration will enable comprehensive patient modeling that guides AI-driven therapeutic design and optimization. Healthcare professionals will leverage AI in augmenting care delivery, allowing provision of safer, standardized, and more effective treatments. Clinicians will utilize 'AI digital consults' to examine digital twin models of patients, testing effectiveness, safety, and patient experience of interventions in digital environments before real-world implementation [Vallée, 2024].

Global Healthcare Transformation

The global transformation of healthcare systems through 2D nanomaterial integration will fundamentally reshape medical practice, healthcare accessibility, and population health outcomes by 2030-2040. This transformation encompasses technological convergence, infrastructure development, and policy frameworks that collectively enable worldwide access to advanced personalized medicine capabilities [BLOG].

Market growth projections indicate substantial expansion in 2D materials markets, with the global market valued at USD 1.85 billion in 2024 and projected to reach USD 16.5 billion by 2032, representing a remarkable 31.7% compound annual growth rate. The healthcare sector is anticipated to exhibit the fastest growth, driven by unique properties of 2D materials enabling breakthroughs in sensitive biosensors, advanced drug delivery systems, and novel diagnostic tools addressing critical unmet medical needs.

Technology democratization will enable widespread access to advanced 2D nanomaterial-based healthcare solutions across diverse global populations. Smart healthcare services will form the backbone of efforts to enhance global health through AI, foundation models, cloud computing, and big data integration. These innovations will permeate every aspect of healthcare, from early disease diagnosis to precise treatment delivery, driving full-scale intelligence and higher efficiency in medical services globally [PNOTE].

Infrastructure transformation will support the deployment of AI-driven 2D nanomaterial therapeutic systems across diverse healthcare settings. By 2040, the NHS and similar healthcare systems will be transformed into "best-of-breed" AI-powered models, revolutionizing healthcare delivery through faster diagnoses, personalized treatments, and improved monitoring capabilities. This transformation will generate substantial economic benefits while establishing global benchmarks for AI-integrated healthcare delivery [BLOG].

The convergence of 2D nanomaterials, precision medicine, and artificial intelligence represents an unprecedented opportunity to transform healthcare delivery fundamentally. Success in realizing this vision will require coordinated efforts among researchers, clinicians, regulatory agencies, and healthcare systems to address current limitations while building the technological and policy frameworks necessary for global implementation. The next decade will be critical in establishing the foundations for this transformation, with 2030-2040 representing the era of fully realized personalized, AI-driven, nanomaterial-enhanced healthcare delivery.

9. Conclusions and Key Takeaways

Summary of Current State-of-the-Art

The field of 2D nanomaterials for biomedical applications has reached a remarkable level of maturity, with unprecedented advances achieved across multiple therapeutic and diagnostic domains during 2020-2025. The current state-of-the-art encompasses a diverse portfolio of materials extending far beyond graphene to include transition metal dichalcogenides (TMDs), MXenes, layered double hydroxides (LDHs), metal-organic frameworks (MOFs), and covalent organic frameworks (COFs), each offering unique properties tailored for specific biomedical applications [Murali, et al, 2021]. Technological sophistication has reached extraordinary levels, with 2D nanomaterials demonstrating exceptional capabilities in targeted drug delivery, photothermal therapy, biosensing, tissue engineering, and diagnostic imaging. The planar topography of these materials confers unique physical, chemical, electronic, and optical properties that enable sophisticated interactions with biological systems through their high surface-to-volume ratios and tunable interfacial chemistry. Recent advances have demonstrated that these materials can achieve 2-3 orders of magnitude improvement in

Clinical translation milestones have been achieved with the first-ever human clinical trial involving controlled inhalation of graphene oxide in healthy volunteers, demonstrating that very pure forms of 2D materials can be developed safely for human exposure. This breakthrough study, conducted by researchers from the Universities of Edinburgh and Manchester, represents a historic moment that validates the safety profile necessary for future clinical applications and provides the foundation for subsequent human studies investigating larger populations and different formulations.

therapeutic effectiveness compared to conventional treatments in certain applications [Ranasinghe, et al, 2022].

Commercial market dynamics reflect the transformative potential of these technologies, with the global nanomedicine market valued at USD 241.82 billion in 2024 and projected to reach USD 570.98 billion by 2032. The 2D materials segment specifically shows remarkable growth, with graphene-based biomedical applications projected to reach USD 1,609.3 million by 2030, representing a 35.1% compound annual growth rate. This explosive market expansion reflects increasing adoption of 2D materials across therapeutic and diagnostic applications driven by their unique properties and demonstrated clinical benefits [NMMS].

Critical Success Factors for Clinical Translation

The successful clinical translation of 2D nanomaterials depends on seven core principles encapsulated in the "DELIVER" framework, which has emerged as the gold standard for advancing preclinical development and enhancing clinical investigation success rates. These principles encompass target product profile definition, essential characterization, lead candidate optimization, intellectual property collaboration, preclinical validation, economic and scalable production design, and regulatory pathway identification [Hu, et al, 2025].

Rigorous safety assessment protocols represent the most critical success factor, as demonstrated by the Manchester graphene trial's success through extensive preclinical safety studies, careful material characterization, and conservative exposure parameters. Comprehensive biocompatibility evaluation must address the unique challenges posed by 2D materials' high surface-to-volume ratios, quantum effects, and ability to cross biological membranes. Long-term toxicological studies and systematic evaluation of biodistribution, accumulation, degradation, and elimination pathways are essential for regulatory approval and clinical success.

Manufacturing standardization and quality control emerge as fundamental requirements for successful translation. The establishment of standardized synthetic protocols ensuring consistent material properties and structural control across production facilities represents a critical barrier that must be overcome. Current challenges include achieving batch-to-batch consistency, maintaining uniform particle size distribution, and ensuring reproducible surface functionalization while scaling production to commercial levels.

Strategic patient stratification utilizing biomarker-guided approaches has proven essential for nanomedicine success. The integration of companion diagnostics with 2D material therapies enables identification of patients most likely to benefit from particular treatments, significantly improving clinical trial outcomes. Advanced molecular profiling approaches that consider immune system status, organ function, and genetic variants affecting nanoparticle metabolism are increasingly important for optimal treatment selection [Abdullah, et al, 2025].

Regulatory pathway optimization requires early engagement with regulatory agencies to mitigate approval risks and identify suitable pathways. The complexity of 2D materials necessitates specialized regulatory approaches that address their unique properties while building upon existing frameworks for nanomedicine evaluation. Successful translation requires comprehensive documentation of manufacturing processes, quality control procedures, and facility validation that meets enhanced GMP requirements.

Recommendations for Researchers and Clinicians

For researchers, the immediate priority must focus on standardized synthetic protocols that ensure consistent material properties and structural control. The development of comprehensive characterization standards addressing lateral size, thickness, surface functionalization, crystal structure, and aggregation behavior is essential for reproducible research and regulatory compliance. Advanced analytical techniques including atomic force microscopy, transmission electron microscopy, X-ray photoelectron spectroscopy, and Raman spectroscopy should be standardized across research institutions [Wu, et al, 2025].

Systematic long-term in vivo evaluations are urgently needed to establish comprehensive biocompatibility profiles for clinical translation. Researchers should prioritize understanding biodistribution patterns, degradation pathways, and long-term accumulation effects through well-designed chronic exposure studies. The development of predictive models based on structure-activity relationships will enable more efficient safety evaluation while reducing reliance on extensive animal testing [Wu, et al, 2025].

Integration of advanced analytical techniques, particularly omics approaches, is essential for better understanding material-cell interactions at the molecular level. Multi-omics data integration utilizing genomics, transcriptomics, proteomics, and metabolomics enables holistic understanding of biological responses that can guide rational material design. AI-enhanced analysis of complex biological datasets can identify patterns and relationships that would be impossible to discover through traditional approaches [Mollaand Bitew, 2024].

For clinicians, the emphasis should be on developing expertise in nanomedicine applications and understanding the unique properties and clinical implications of 2D materials. Medical professionals require specialized education on nanomaterial properties, administration techniques, monitoring requirements, and adverse event management. This training represents an ongoing investment that healthcare systems must prioritize to enable successful clinical implementation [Tang, et al, 2021].

Patient selection strategies utilizing precision medicine approaches should be developed to optimize treatment outcomes. Clinicians should work with researchers to identify appropriate biomarkers and companion diagnostics that can guide treatment selection. Understanding individual patient factors affecting nanoparticle behavior, including immune system status and genetic variants affecting metabolism, is crucial for treatment optimization.

Infrastructure adaptation within healthcare systems is necessary to support 2D nanomaterial applications. Specialized storage and handling facilities, advanced monitoring capabilities, and appropriate distribution systems represent significant investments that healthcare institutions must plan for systematically.

Final Thoughts on Transformative Potential

The transformative potential of 2D nanomaterials in healthcare represents one of the most significant paradigm shifts in modern medicine, with implications extending far beyond incremental improvements to fundamentally reshape therapeutic approaches. These materials offer unprecedented opportunities to address complex medical challenges through their unique combination of atomic-thin structure, large surface area, tunable properties, and sophisticated biological interactions.

Revolutionary therapeutic capabilities enabled by 2D materials include the development of smart responsive systems that can autonomously monitor physiological conditions, make therapeutic decisions, and adjust treatment parameters in real-time. The integration of artificial intelligence with 2D material platforms creates opportunities for predictive therapeutic systems that can anticipate disease progression and optimize interventions before clinical symptoms develop. This represents a fundamental shift from reactive to proactive healthcare delivery.

Precision medicine transformation through 2D materials enables truly personalized therapeutic approaches that account for individual genetic profiles, molecular signatures, and environmental factors. The ability to design patient-specific nanomaterial platforms tailored to individual biology represents the ultimate realization of precision medicine principles. Digital twin technologies combined with 2D materials will enable testing of therapeutic interventions in virtual patient models before clinical implementation, optimizing treatment strategies while minimizing risks [NOVUM].

Global healthcare accessibility stands to be revolutionized through 2D material technologies, with market projections indicating widespread democratization of advanced medical capabilities. The integration of 2D materials with telemedicine platforms, wearable monitoring systems, and point-of-care diagnostic devices will enable delivery of sophisticated healthcare services to underserved populations globally. This technological convergence has the potential to address healthcare disparities and improve population health outcomes worldwide.

The future vision for 2030-2040 encompasses fully integrated healthcare ecosystems where 2D nanomaterials serve as the foundational technology platform for AI-driven, personalized, and continuously adaptive medical care. This transformation will require unprecedented collaboration among researchers, clinicians, regulatory agencies, industry partners, and healthcare systems to realize the full potential of these remarkable materials [Asma, 2025].

The journey toward clinical implementation remains challenging, but the foundational science, regulatory frameworks, and commercial investments are aligning to support widespread adoption. Success will depend on maintaining rigorous safety standards, developing robust manufacturing processes, and ensuring equitable access to these transformative technologies. The next decade represents a critical period for establishing the infrastructure and capabilities necessary to deliver on the extraordinary promise of 2D nanomaterials in revolutionizing healthcare delivery for global populations.

References:

- Abdullah, K.M., Sharma, G., Singh, A.P., et al. Nanomedicine in Cancer Therapeutics: Current Perspectives from Bench to Bedside. Molecular cancer 2025 24(1) 169
- Abou Aitah, K., Sabbagh, F., and Kim, B.S. Graphene Oxide Nanostructures as Nanoplatforms for Delivering Natural Therapeutic Agents:
 Applications in Cancer Treatment, Bacterial Infections, and Bone Regeneration Medicine. Nanomaterials (Basel, Switzerland), 2023, 13(19), 2666.
- Agema, B.C., Koch, B.C.P., Mathijssen, R.H.J., et al. From Prospective Evaluation to Practice: Model-Informed Dose Optimization in Oncology. *Drugs*, 2025,85(4), 487–503.
- Ahmad, R.M., and Mohammad, M.S.Towards the use of nanotechnology and pharmacogenomics in personalized medicine. Int. J. Advances in Engineering and Management (IJAEM), 2021, 3, 12, 689-692.
- Ahmed, H., May 2022; The Challenges Behind Scaling Up Nanomaterials. Text @ https://www.azonano.com/article.aspx?ArticleID=6126
- Aljabali, A.A.A., Obeid, M.A., Gammoh, O., et al. Nanomaterial-Driven Precision Immunomodulation: A New Paradigm in Therapeutic Interventions. Cancers, 2024, 16(11), 2030.
- Andrews, J.P.M., Joshi, S.S., Tzolos, E. et al. First-in-human controlled inhalation of thin graphene oxide nanosheets to study acute cardiorespiratory responses. Nat. Nanotechnol., 2024, 19, 705–714.
- Antoniou, M., Kolamunnage-Dona, R., Wason, J., et al. Biomarker-guided trials: Challenges in practice. Contemporary clinical trials communications, 2019, 16, 100493.
- ARTICLE-01; Understanding the degradation of 2D materials: Implications for human health and environmental safety assessment. Text @ rivm.nl/en/weblog/understanding-degradation-of-2d-materials-implications-for-human-health-and-environmental
- ARTICLE-G; 2D Materials Innovation for Biomedical Applications. Text @ https://graphene-flagship.eu/materials/news/2d-materials-innovation-for-biomedical-applications/

- Arun, S., Sykes, E.R., and Tanbeer, S. RemoteHealthConnect: Innovating patient monitoring with wearable technology and custom visualization. *Digital health*, 2024,10, 20552076241300748.
- Asma, A.N. AI and Healthcare in 2030: Predictions and Pathways. J. AI-Powered Medical Innovations (International Online ISSN 3078-1930), 2025, 3(1), 84-101.
- Bai, M., Wan, H., Zhang, Y., et al. Two-dimensional nanomaterials based on rare earth elements for biomedical applications. Chem. Sci., 2024,15, 16887-16907
- Baig, N. Two-dimensional nanomaterials: A critical review of recent progress, properties, applications, and future directions, Composites Part A: Applied Science and Manufacturing, 2023, 165, 107362.
- Bartolomucci, A., Nobrega, M., Ferrier, T. et al. Circulating tumor DNA to monitor treatment response in solid tumors and advance precision oncology. npj Precis. Onc., 2025, 9, 84.
- BLOG-01; The Fusion of Precision Drug Delivery and Personalized Medicine. Text @ https://xtpl.com/the-fusion-of-precision-drug-delivery-and-personalized-medicine/
- BLOG; The AI Revolution in Healthcare: A Consolidated 2040 Vision for the NHS. Text @ https://dralfoldman.com/2025/02/07/the-ai-revolution-in-healthcare-a-consolidated-2040-vision-for-the-nhs/
- Bolotsky, A.,Butler, D.,Dong, C., et al. Two-Dimensional Materials in Biosensing and Healthcare: From In Vitro Diagnostics to Optogenetics and Beyond. ACS Nano, 2019, 13, 9, 9781–9810
- Bongurala, A.R., Save, D., Virmani, A., et al. Progressive role of artificial intelligence in treatment decision-making in the field of medical oncology. Front. Med., 13 February 2025, Sec. Precision Medicine, 12 2025, doi.org/10.3389/fmed.2025.1533910
- Campos, M.T., Silva, F.A.L.S., Fernandes, J.R., et al. New MoS₂/Tegafur-Containing Pharmaceutical Formulations for Selective LED-Based Skin Cancer Photo-Chemotherapy. *Pharmaceutics*, 2024, 16(3), 360.
- Cao, J., Qi, J., Lin, X., Xiong, Y., He, F., Deng, W., & Liu, G. Biomimetic Black Phosphorus Nanosheet-Based Drug Delivery System for Targeted Photothermal-Chemo Cancer Therapy. Frontiers in bioengineering and biotechnology, 2021, 9, 707208.
- Chen, X., Wu, D., and Chen, Z. Biomedical applications of stimuli-responsive nanomaterials. MedComm, 2024, 5(8), e643.
- CLINTRIAL; FIH Clinical Investigation of Graphene Electrodes for Brain Mapping; Clinical Trials Gov. ID: NCT06368310;Sponsor: University of Manchester,Information provided Coope, D. Northern Care Alliance NHS Foundation Trust. Text @ clinicaltrials. gov/study/NCT06368310?term=graphene&rank=4&checkSpell=
- Costa, A., and Jornitz, M. November 14, **2024**. The New Continuous and Controlled Approach to Nanoparticle Processing. Text @bioprocessintl.com/continuous-bioprocessing/the-new-continuous-and-controlled-approach-to-nanoparticle-processing
- Dai, W. Two-Dimensional Materials in Nanomaterials: Properties, Applications, and Prospects. Proceedings of the 2nd International Conference on Functional Materials and Civil Engineering DOI: 10.54254/2755-2721/89/20241095
- Davis, R., Jr, Urbanowski, R.A., Jr, and Gaharwar, A.K. 2D layered nanomaterials for therapeutics delivery. Current opinion in biomedical engineering, 2021, 20, 100319.
- de Lima, B.P., Ferraz, L.S., Devalle, S., et al. Liquid Biopsy-Derived Tumor Biomarkers for Clinical Applications in Glioblastoma. *Biomolecules*, 2025, 15(5), 658.
- Demuth, S., De Sèze, J., Edan, G., et al. Digital Representation of Patients as Medical Digital Twins: Data-Centric Viewpoint, JMIR Med. Inform.,
 2025. 13. e53542.
- Dharani, S., and Kamraj, R. A Review of the Regulatory Challenges of Personalized Medicine. Cureus, 2024, 16(8), e67891.
- Dixon, D., Sattar, H., Moros, N., et al. Unveiling the Influence of AI Predictive Analytics on Patient Outcomes: A Comprehensive Narrative Review. *Cureus*, **2024**, *16*(5), e59954.
- Dubey, R. Cowles, M., Salimi, Z., et al. Boron nitride nanosheets, quantum dots, and dots: Synthesis, properties, and biomedical applications.
 Special Collection: 2D Materials for Biomedical Applications. APL Mater., 2025, 13, 040601.
- Durgadevi, S., and Kumar, P. Surface-tailored graphene nanosheets targeting PI3K/Akt signaling of breast cancer cells. Scientific reports, 2025, 15(1), 29737.
- Elrashedy, A., Mousa, W., Nayel, M., et al. Advances in bioinformatics and multi-omics integration: transforming viral infectious disease research in veterinary medicine. Virology journal, 2025, 22(1), 22.
- Eshghi, S., and Jookar Kashi, F. Acid Red 88 biodegradation by Cu nanoparticles stabilized on Marinospirillumalkaliphilum strain N. Sci Rep., 2025, 15, 18903.
- Fang, G., Annis, I.E., Elston-Lafata, J., et al. Applying machine learning to predict real-world individual treatment effects: insights from a virtual patient cohort. *Journal of the American Medical Informatics Association: JAMIA*, 2019, 26(10), 977–988.
- FDA; Medical Device Material Safety Summaries; Text @ https://www.fda.gov/medical-devices/science-and-research-medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/medical-devices/me
- FHTRIAL; 16 February 2024; First human trial shows 'wonder' material can be developed safely. Text @ https://www.manchester.ac.uk/about/news/first-human-trial-shows-wonder-material-can-be-developed-safely/
- FMI-01, At-Home Testing Market Analysis Growth, Demand & Forecast 2025 to 2035. Text @ https://www.futuremarketinsights.com/reports/at-home-testing-market
- Fusco, L., Gazzi, A., Peng, G., et al. Graphene and other 2D materials: a multidisciplinary analysis to uncover the hidden potential as cancer theranostics. *Theranostics*, 2020, 10(12), 5435-5488.
- Gawne, P.J., Ferreira, M., Papaluca, M., et al. New Opportunities and Old Challenges in the Clinical translation of Nanotheranostics. *Nature reviews. Materials*, 2023, 8(12), 783–798.

- GMS; Graphene Market Size, Share & Trends Analysis Report. Text @https://www.grandviewresearch.com/industry-analysis/graphene-industry
- Goetz, L.H., and Schork, N.J. Personalized medicine: motivation, challenges, and progress. Fertility and sterility, 2018, 109(6), 952-963.
- Gomez, L., Sebastian, V., Irusta, S., et al. Scaled-up production of plasmonic nanoparticles using microfluidics: from metal precursors to functionalized and sterilized nanoparticles_Lab Chip, 2014,14, 325-332
- Gravagnuolo, A.M., Martucci, A., and Morales-Naváez, E. Editorial: Biointerfacing 2D nanomaterials and engineered heterostructures, volume II. Frontiers in bioengineering and biotechnology, 2024, 12, 1542498.
- GSTOCK; Top Graphene Stocks 2025: Pure-Play Watchlist. Last modified: May 20, 2025. Text @ https://exoswan.com/graphene-stocks
- Guiney, L.M., Wang, X., Xia, T., et al. Assessing and Mitigating the Hazard Potential of Two-Dimensional Materials. ACS nano, 2018, 12(7), 6360–6377.
- Halamoda-Kenzaoui, B., and Bremer-Hoffmann, S. Main trends of immune effects triggered by nanomedicines in preclinical studies. Int J. Nanomedicine. 2018, 13, 5419-5431.
- Hassan M, Liu S, Liang Z, et al. Revisiting traditional and modern trends in versatile 2D nanomaterials: Synthetic strategies, structural stability, and gas-sensing fundamentals. *Journal of Advanced Ceramics*, 2023, 12(12): 2149-2246.
- Hassan, S., Prakash, G., Ozturk, A., et al. 2017). Evolution and Clinical Translation of Drug Delivery Nanomaterials. Nano today, 2017, 15, 91–106
- Holca, A., Cucuiet, V., Astilean, S., et al. Recent advances in gold nanoparticle-graphene hybrid nanoplatforms with visible to near-infrared response for photodynamic and photothermal therapy and bioimaging. RSC advances, 2025,15(15), 11902–11922.
- Hu, J., and Dong, M. Recent advances in two-dimensional nanomaterials for sustainable wearable electronic devices. *Journal of nanobiotechnology*, 2024, 22(1), 63.
- Hu, K., Xie, L., Zhang, Y. *et al.* Marriage of black phosphorus and Cu²⁺ as effective photothermal agents for PET-guided combination cancer therapy. *Nat Commun.*, *2020*, 11, 2778 (2020).
- Ikwelle, T.A., Ihim, A.C., Ozuruoke, D.F., et al. Multi-Omics Integration in Personalized Medicine: Advancing Laboratory Diagnostics and Precision Therapeutics in the Era of Individualized Healthcare. J. Drug Delivery Ther., 2025, 15(5),132-4.
- INDBRIEF; Rao, A., February 13, 2025; Pharmaceutical SMEs in India Gain Time to Meet Enhanced GMP Standards Under Revised Schedule
 M. Text @ https://www.india-briefing.com/news/india-extends-gmp-compliance-deadline-to-december-2025-35679.html/
- INDMS; India Graphene Market Size, Share, Trends and Forecast by Type, Application, End-Use Industry, and Region, 2025-2033; Report ID: SR112025A29472; Text @ https://www.imarcgroup.com/india-graphene-market
- INVEST; Pistilli, M., Mar. 17, 2025. Investing in Graphene Companies. Text @ https://investingnews.com/daily/tech-investing/nanoscience-investing/graphene-investing/investing-in-graphene-companies/
- Johnson, K.B., Wei, W.Q., Weeraratne, D., et al. Precision Medicine, AI, and the Future of Personalized Health Care. *Clinical and translational science*, **2021**,*14*(1), 86–93.
- Katsoulakis, E., Wang, Q., Wu, H. et al. Digital twins for health: a scoping review. npj Digit. Med., 2024, 7, 77.
- Kaur, A., Babaliari, E., Bolanos-Garcia, V.M. et al. Assessment of aqueous graphene as a cancer therapeutics delivery system. Sci Rep., 2025, 15, 15396.
- Khafaga, D.S.R., El-Morsy, M.T., Faried, H., et al. Metal-organic frameworks in drug delivery: engineering versatile platforms for therapeutic applications. RSC advances, 2024, 14(41), 30201–30229.
- Khakpour, E., Salehi, S., Naghib, S. M., et al. Graphene-based nanomaterials for stimuli-sensitive controlled delivery of therapeutic molecules. Frontiers in bioengineering and biotechnology, 2023, 11, 1129768.
- Kim, S., Padmanaban, S., Sundaram, A., et al. Optimizing the Surface Functionalization of Peptide-MXene Nanoplatforms to Amplify Tumor-Targeting Efficiency and Photothermal Therapy. Biomaterials research, 2025, 29, 0198.
- Kumar, R.,Singh, S.K.,Srivastava, R.K., et al. NIR-Responsive Black Phosphorus Nanosheet-Integrated Niosomes for Combinatorial Chemophototherapy of Cancers. ACS Bio Med Chem Au., 2025, 5, 1, 143–153
- Kyriakides, T.R., Raj, A., Tseng, T.H., et al. Biocompatibility of nanomaterials and their immunological properties. Biomedical materials (Bristol, England), 2021, 16(4), 10.1088/1748-605X/abe5fa.
- Lee, S.June 26, 2025, Nanomedicine in Pharmacogenomics: A New Era. Text @ https://www.numberanalytics.com/blog/nanomedicine-pharmacogenomics-ultimate-guide
- Li, Z., Weller, C.A., Shah, S., et al. ProtPipe: A Multifunctional Data Analysis Pipeline for Proteomics and Peptidomics. Genomics, Proteomics & Bioinformatics, 2024, 22, 6, gzae083.
- Li, K., Ji, Q., Liang, H., et al. Biomedical application of 2D nanomaterials in neuroscience. *Journal of nanobiotechnology*, **2023**, 21(1), 181.
- Mahajan, A., and Powell, D. Generalist medical AI reimbursement challenges and opportunities. NPJ digital medicine, 2025, 8(1), 125.
- Malik, S., Muhammad, K., and Waheed, Y. Emerging Applications of Nanotechnology in Healthcare and Medicine. Molecules (Basel, Switzerland), 2023, 28(18), 6624.
- Mani, S., Lalani, S.R. and Pammi, M. Genomics and multiomics in the age of precision medicine. *Pediatr Res.*, 2025, 97, 1399–1410 (2025).
- Marques, L., and Vale, N. Toward Personalized Salbutamol Therapy: Validating Virtual Patient-Derived Population Pharmacokinetic Model with Real-World Data. Pharmaceutics, 2024, 16(7), 881.
- Mathew, M., and Rout, C.S. Electrochemical biosensors based on Ti₃C₂T_x MXene: future perspectives for on-site analysis. Current Opinion in Electrochemistry, 2021, 30, 100782.
- Molla, G., and Bitew, M. Revolutionizing Personalized Medicine: Synergy with Multi-Omics Data Generation, Main Hurdles, and Future Perspectives. Biomedicines, 2024, 12(12), 2750.

- MR-01; Artificial Intelligence In Drug Discovery Market Summary. Text @ https://www.grandviewresearch.com/industry-analysis/artificial-intelligence-drug-discovery-market
- MR-02; Home Diagnostics Market Size and Trends. Text @ novaoneadvisor. com/report/home-diagnostics-market
- Mukherjee, S., Honkala, A., and Rolnik, B. Review of the Clinical Trial Failures of Nanomedicines Targeting Cancer. *J. Student Research*, **2022**, *11*(2). https://doi.org/10.47611/jsrhs.v11i2.2607
- Murali, A., Lokhande, G., Deo, K.A., et al. Emerging 2D Nanomaterials for Biomedical Applications. *Materials today (Kidlington, England)*, 2021, 50, 276–302.
- Nakamura, Y., Ozaki, H., Ueno, M. et al. Targeted therapy guided by circulating tumor DNA analysis in advanced gastrointestinal tumors. Nat Med., 2025, 31, 165–175.).
- Nishida, A., Sawada, Y., Arai, R., et al. Evaluation of the immunotoxicity potential of nanomaterials using THP-1 cells. Front. Toxicol., 01 July **2024**; Sec. Nanotoxicology; 6 **2024**; doi.org/10.3389/ftox.2024.129314
- NMCOM; Top 20 Global Companies in Nanomedicine Market (2025-2035): Competitive Analysis and Forecast. Aug 2025 Author, Spherical Insights. Text
 https://www.sphericalinsights.com/blogs/top-20-global-companies-in-nanomedicine-market-2025-2035-competitive-analysis-and-forecast
- NMMR-01; Nanomedicine Market: Global Industry Analysis and Forecast (2024-2030). Text @ maximizemarketresearch.com/market-report/nanomedicine-market/39223/
- NMMS; Nanomedicine Market Size; Text @ fortunebusinessinsights. com/nanomedicine-market-110983
- NNOTE-01; Nanotechnology-based medicinal products for human use EU-IN Horizon Scanning Report January, 2025, EMA/20989/2025. Text
 @ https://www.ema.europa.eu/en/documents/report/nanotechnology-based-medicinal-products-human-use-eu-horizon-scanning-report_en.pdf
- NOVUM; Unveiling the Future of Personalized Medicine. Text @novumgen.com/unveiling-the-future-of-personalized-medicine.html
- Oei, S.P., Bakkes, T.H.G.F., Mischi, M., et al. Artificial intelligence in clinical decision support and the prediction of adverse events. *Frontiers in digital health*, **2025**, *7*, 1403047.
- Padamati, M., and Rangnath, M.M. Regulatory Considerations for Personalized Medicine and Precision Therapeutics. Int. J. Trend in Scientific Research and Development (IJTSRD), 2023, 7(6), 47-52.
- Patne, A.Y., Mohapatra, S., and Mohapatra, S.S. Role of Nanomedicine in Transforming Pharmacotherapy for Substance Use Disorder (SUD). Wiley interdisciplinary reviews. Nanomedicine and nanobiotechnology, 2025, 17(2), e70008.
- PNOTE-02; Unveiling the Future of Personalized Medicine. Text @ https://www.novumgen.com/unveiling-the-future-of-personalized-medicine.html
- PNOTE; Intelligent World 2030; Text @ https://www-file.huawei.com/-/media/corp2020/pdf/giv/2024/intelligent_world_2030_2024_en.pdf
- POCS-01; Point of Care (PoC) Diagnostic Solutions. Text @ analog.com/en/solutions/healthcare/in-vitro-diagnostics/point-of-care-poc-diagnostics.html
- PROD-01; Using Flow Technology in Nanoparticle Synthesis. Text @ https://ardena.com/resources/using-flow-technology-in-nanoparticle-synthesis/
- Qamar, W., Gulia, S., Athar, M., et al. An insight into impact of nanomaterials toxicity on human health. Peer J., 2024, 12, e17807.
- Ranasinghe, J.C., Jain, A., Wu, W., et al. Engineered 2D materials for optical bioimaging and path toward therapy and tissue engineering. *Journal of materials research*, 2022, 37(10), 1689–1713.
- Rehan, F., Zhang, M., Fang, J., et al. Therapeutic Applications of Nanomedicine: Recent Developments and Future Perspectives. Molecules (Basel, Switzerland), 2024, 29(9), 2073.
- Remya, R.R., Julius, A., Suman, T.Y., et al. Role of Nanoparticles in Biodegradation and Their Importance in Environmental and Biomedical Applications. Hindawi, Journal of Nanomaterials. 2022, Article ID 6090846
- Rodríguez-Gómez, F.D., Monferrer, D., Penon, O., et al. Regulatory pathways and guidelines for nanotechnology-enabled health products: a comparative review of EU and US frameworks. Front. Med., 05 March 2025, Sec. Regulatory Science, 12 2025; doi.org/10.3389/fmed.2025.1544393
- RPMS-01; Remote Patient Monitoring System Market Summary; Text @ https://www.grandviewresearch.com/industry-analysis/remote-patient-monitoring-devices-market
- Sanità, G., Carrese, B., and Lamberti, A. Nanoparticle Surface Functionalization: How to Improve Biocompatibility and Cellular Internalization. Frontiers in molecular biosciences, 2020, 7, 587012.
- Saratkar, S.Y., Langote, M., Kumar, P., et al. Digital twin for personalized medicine development. Front. Digit. Health, 07 August 2025, Sec. Personalized Medicine, 7 2025, doi.org/10.3389/fdgth.2025.1583466
- Sattari, S., Adeli, M., Beyranvand, S., et al. Functionalized Graphene Platforms for Anticancer Drug Delivery. Int. j. nanomedicine, 2021, 16, 5955–5980.
- Sengupta, J., and Hussain, C.M. MXene-Based Electrochemical Biosensors: Advancing Detection Strategies for Biosensing (2020-2024). *Biosensors*, **2025**, *15*(3), 127.
- Serrano, D.R., Luciano, F.C., Anaya, B.J., et al. Artificial Intelligence (AI) Applications in Drug Discovery and Drug Delivery: Revolutionizing Personalized Medicine. Pharmaceutics, **2024**, 16(10), 1328.
- Sharma, H., Bhadouria, U., Sharma, T., et al.An Updated Overview on Personalized Medicine: The Next-Gen Paradigm. IJPQA, 2023, 14, 2, 456-463.
- Shen, H., Zhang, L., Liu, M., et al. Biomedical applications of graphene. *Theranostics*, 2012, 2(3), 283–294.

- Sun, D., Gao, W., Hu, H., et al. Why 90% of clinical drug development fails and how to improve it?. Acta pharmaceutica Sinica. B, 2022, 12(7), 3049–3062.
- Tang, Z., Xiao, Y., Kong, N., et al. Nano-bio interfaces effect of two-dimensional nanomaterials and their applications in cancer immunotherapy. Acta pharmaceutica Sinica. B, 2021, 11(11), 3447–3464.
- Tanveer, H., Faheem, M., Khan, A.H., et al. Machine Learning Models for Personalized Treatment Recommendations in Diabetes Management. *Int. J. Scientific Research and Management (IJSRM)*, **2025**, *13*(04), 2145-2154.
- Tenchov, R., Hughes, K.J., Ganesan, M., et al. Transforming Medicine: Cutting-Edge Applications of Nanoscale Materials in Drug Delivery. ACS nano, 2025, 19(4), 4011–4038.
- Tiwari, R., Dev, D., Thalla, M., et al. Nano-enabled pharmacogenomics: revolutionizing personalized drug therapy. *Journal of biomaterials science*. *Polymer edition*, **2025**, *36*(7), 913–938.
- USFDA; Current Good Manufacturing Practice (CGMP) Regulations. Text @ https://www.fda.gov/drugs/pharmaceutical-quality-resources/current-good-manufacturing-practice-cgmp-regulations
- Uzdrowska, K., Knap, N., Gulczynski, J., et al. Chasing Graphene-Based Anticancer Drugs: Where are We Now on the Biomedical Graphene Roadmap?. International journal of nanomedicine, 2024, 19, 3973

 –3989.
- Vaghasiya, J.V., Mayorga-Martinez, C.C. and Pumera, M. Wearable sensors for telehealth based on emerging materials and nanoarchitectonics. npj Flex Electron., 2023, 7, 26.
- Vallée, A. Envisioning the Future of Personalized Medicine: Role and Realities of Digital Twins. J. Med. Internet. Res., 2024, 26, e50204
- van der Meel, R., Sulheim, E., Shi, Y. et al. Smart cancer nanomedicine. Nat. Nanotechnol., 2019, 14, 1007–1017.
- VMR; Global 2D Carbon Material Graphene Market Size. Text @ verifiedmarketreports.com/product/2d-carbon-material-graphene-market/
- Wang, R.S., Maron, B.A., and Loscalzo, J. Multiomics Network Medicine Approaches to Precision Medicine and Therapeutics in Cardiovascular Diseases. Arteriosclerosis, thrombosis, and vascular biology, 2023, 43(4), 493–503.
- Wang, Y., Li, T., Li, Y., et al. 2D-Materials-Based Wearable Biosensor Systems. *Biosensors*, 2022, 12(11), 936.
- Wu, G., Yang, B.B., and Yang, Y.-W. Two-dimensional (2D) materials for biomedical applications. APL Mater., 2025, 13, 030401.
- Yadav, P.K., Kumar, P., Teja, D.R., et al. A Review on Low-Dimensional Nanomaterials: Nanofabrication, Characterization and Applications. *Nanomaterials (Basel, Switzerland)*, **2022**, *13*(1), 160.
- Yang, X., Zhang, C., Deng, D., et al. Multiple Stimuli-Responsive MXene-Based Hydrogel as Intelligent Drug Delivery Carriers for Deep Chronic Wound Healing. Small (Weinheim an der Bergstrasse, Germany), 2022, 18(5), e2104368.
- Zhang, H. Ultrathin Two-Dimensional Nanomaterials. ACS Nano, 2015, 9, 10, 9451–9469
- Zhang, H., Li, S., and Ma, X. Transforming Healthcare with Nanomedicine: A SWOT Analysis of Drug Delivery Innovation. Drug design, development and therapy, 2024, 18, 3499–3521.
- Zhang, L., You, J., Lv, H., et al. Black Phosphorus A Rising Star in the Antibacterial Materials. *International journal of nanomedicine*, 2023, 18, 6563–6584.
- Zhang, R., Yan, Z., Gao, M., et al. Recent advances in two-dimensional materials for drug delivery. J. Mater. Chem. B, 2024, 12, 12437-12469
- Zhao, J., Li, T., Yue, Y., et al. Advancements in employing two-dimensional nanomaterials for enhancing skin wound healing: a review of current
 practice. Journal of nanobiotechnology, 2024, 22(1), 520.
- Zheng, C., Li, M., and Ding, J. Challenges and Opportunities of Nanomedicines in Clinical Translation. BIO Integration, 2021, 2(2), 57–60.
- Zhu, Z., Ren, G., Xiong, S., et al. Rare earth-based two-dimensional materials for biomedical research. J. Mater. Chem. B, 2025, 13, 9679-9697.